

<mark>_</mark>

JGR Solid Earth

RESEARCH ARTICLE

10.1029/2023JB027260

Key Points:

- We provide a complete catalog of high-frequency rupture histories for $M \ge 7.5$ events 2010–2022
- We develop a semi-automatic method for estimating rupture length, speed, directivity, and aspect ratio
- Both encircling ruptures and emissions updip of slip asperities common in megathrust earthquakes

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

F. Vera, fvera@gfz-potsdam.de

Citation:

Vera, F., Tilmann, F., & Saul, J. (2024). A decade of short-period earthquake rupture histories from multi-array back-projection. *Journal of Geophysical Research: Solid Earth*, *129*, e2023JB027260. https://doi.org/10.1029/2023JB027260

Received 16 JUN 2023 Accepted 18 JAN 2024

Author Contributions:

Investigation: Felipe Vera Supervision: Frederik Tilmann, Joachim Saul Writing – original draft: Felipe Vera Writing – review & editing: Frederik Tilmann, Joachim Saul

© 2024. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

A Decade of Short-Period Earthquake Rupture Histories From Multi-Array Back-Projection

Felipe Vera^{1,2}, Frederik Tilmann^{1,2}, and Joachim Saul¹

¹German Research Centre for Geosciences GFZ, Potsdam, Germany, ²Institute for Geological Sciences, Freie Universität Berlin, Berlin, Germany

Abstract Teleseismic back-projection imaging has emerged as a powerful tool for understanding the rupture propagation of large earthquakes. However, its application often suffers from artifacts related to the receiver array geometry. We developed a teleseismic back-projection technique that can accommodate data from multiple arrays. Combined processing of P and pP waveforms may further improve the resolution. The method is suitable for defining arrays ad-hoc to achieve a good azimuthal distribution for most earthquakes. We present a catalog of short-period rupture histories (0.5–2.0 Hz) for all earthquakes from 2010 to 2022 with $M_W \ge 7.5$ and depth less than 200 km (56 events). The method provides automatic estimates of rupture length, directivity, speed, and aspect ratio, a proxy for rupture complexity. We obtained short-period rupture length scaling relations that are in good agreement with previously published relations based on estimates of total slip. Rupture speeds were consistently in the sub-Rayleigh regime for thrust and normal earthquakes, whereas a tenth of strike-slip events propagated at supershear speeds. Many rupture histories exhibited complex behaviors, for example, rupture on conjugate faults, bilateral propagation, and dynamic triggering by a P wave. For megathrust earthquakes, ruptures encircling asperities were frequently observed, with downdip, updip, and balanced patterns. Although there is a preference for short-period emissions to emanate from central and downdip parts of the megathrust, emissions updip of the main asperity are more frequent than suggested by earlier results.

Plain Language Summary Back-projection is an earthquake imaging method based on seismic waveforms recorded remotely at a group of seismometers (seismic array). Here, we develop a new approach to combine backprojections from multiple arrays and seismic waveforms and use it to derive a catalog of large earthquake rupture histories from 2010 to 2022, providing a map view of the high-frequency radiation emitted along the fault. The method automatically estimates the earthquake rupture length, speed, directivity, and aspect ratio. Based on these estimates, we obtained scaling relations between the earthquake magnitude and rupture length that agree with classical relationships. We identified strike-slip earthquakes propagating at supershear, that is, faster than the shear wave speed, the usual limit for self-sustaining rupture propagation. We observed complex rupture behaviors, for example, multiple faults activated, bilateral ruptures, and triggering of the main phase of a rupture by a primary (P) wave from the earliest part of the rupture. For subduction earthquakes, high-frequency emissions were often observed, forming a ring around the fault interface patches (asperities) where the main slip occurs. There was a preference for high-frequency radiation to emanate from central and deeper parts of the subducting plate interface, but shallower emissions were more frequent than expected from previous literature.

1. Introduction

Following the extraordinary 26 December 2004 Sumatra-Andaman (M_W 9.3) earthquake, back-projection rupture imaging has become an important technique to complement finite-fault source inversions for determining the kinematics of very large ruptures (e.g., Ishii et al., 2005; Krüger & Ohrnberger, 2005; Walker et al., 2005). Backprojection exploits the coherence of (usually) teleseismic P waveforms with limited prior assumptions on the fault geometry. Applications have targeted, for example, megathrust subduction earthquakes (e.g., Koper et al., 2011; Lay et al., 2010; Meng et al., 2011; Palo et al., 2014), aftershock detection (e.g., Feng et al., 2020; Kiser & Ishii, 2013; Tilmann et al., 2016), complex ruptures along multiple faults (e.g., Lay et al., 2018; Meng, Ampuero, Stock, et al., 2012; Ruppert et al., 2018), intermediate-depth earthquakes (e.g., Kiser et al., 2011; Ye et al., 2020), moderate size earthquakes (e.g., D'Amico et al., 2010; Taymaz et al., 2021), and near-field tsunami prediction (e. g., An & Meng, 2016; Xie & Meng, 2020), among others.



Back-projection is generally applied to earthquakes that are large enough to reveal source complexity in teleseismic recordings. The technique makes use of the coherency of short-period filtered waveforms, which are able to resolve details of the earthquake rupture better than long-period waveforms (e.g., Kiser & Ishii, 2017). The frequency band used must subsequently be taken into account in the interpretation of the inferred rupture history. Short-period radiation is related to variations of slip and rupture velocity (e.g., Madariaga, 1977; Madariaga, 1983; Marty et al., 2019). This implies that the back-projected rupture history can be related to fault heterogeneities. For example, subduction zone megathrust earthquakes were found to radiate short-period energy predominantly in deeper parts of the megathrust, downdip of the maximum slip regions derived from finite-fault inversions, which are sensitive to the low-frequency emissions of the rupture (e.g., Lay et al., 2012; Yao et al., 2011, 2013). The apparently dominant downdip short-period radiation is proposed to occur at the transition between brittle and ductile regions (e.g., Lay et al., 2012; Simons et al., 2011). Recently, K. Wang et al. (2020) correlated the 27 February 2010 Maule ($M_W 8.8$) earthquake short-period rupture with downdip segmentation at the base of the overriding mantle wedge. In contrast to this view of the along-dip segmentation of seismic emission during large earthquakes, Meng et al. (2018) provided the first evidence for an encircling rupture, that is, high-frequency rupture emissions both updip and downdip of the slip patch for the 16 September 2015 Illapel (M_{W} 8.3) earthquake. Similarly, Vera et al. (2023) presented rupture emissions outlining the main slip asperities (updip and downdip) for the 14 November 2007 Tocopilla (M_W 7.7) earthquake. Moreover, the observations of Meng et al. (2018) for the 2015 Illapel event are in contrast to earlier back-projection studies, which had shown the expected, predominantly downdip rupture pattern (e.g., Melgar et al., 2016; Tilmann et al., 2016; Yin et al., 2016). It raises the question if this type of complex behavior is actually more common and just requires higher resolution rupture images.

Fracture mechanics predicts that instabilities such as earthquakes either propagate at sub-Rayleigh speed or exceed the shear wave speed (e.g., Andrews, 1976; Burridge, 1973; Das & Aki, 1977), where the latter is only expected for mode II cracks. In mode II cracks, the rupture propagates in the direction of displacement, which is also the direction of the initial shear stress resolved onto the fault. Such supershear earthquakes are associated with simple fault geometry and homogeneous stress-strength conditions (e.g., Bouchon et al., 2010), but also to damage zones under relatively low stress leading to unstable supershear ruptures, that is, between the shear wave speed and $\sqrt{2}$ times the shear wave speed (e.g., Burridge et al., 1979). Many strike-slip earthquakes with supershear speeds have been reported in the literature, see Robinson et al. (2010) for a review, for example, the 1979 Imperial Valley (e.g., Archuleta, 1984; Spudich & Cranswick, 1984), 1999 Izmit (e.g., Bouchon et al., 2001), 2002 Denali (e.g., Dunham & Archuleta, 2004; Walker & Shearer, 2009), 2001 Kunlunshan (e.g., Bouchon & Vallée, 2003), 2013 Craig (e.g., Aderhold & Abercrombie, 2015; Yue et al., 2013) and the 2018 Palu earthquake (e.g., Bao et al., 2019; Socquet et al., 2019).

A disadvantage of the back-projection method is that the array configuration can cause notable artifacts in the recovered rupture (e.g., Meng, Ampuero, Luo, et al., 2012). Back-projection often leads to a persistent time-space trade-off of the earthquake rupture toward the seismic array, often called "swimming" artifact. This effect arises from the low curvature of the time-distance travel time curve. Sources closer to the receivers but activated later will have the same arrival time as earlier sources farther away, resulting in a point source in space and time appearing as an extended source drifting toward the array. In animations showing the evolution of the back-projected energy with time, this slightly irregular drifting looks like a swimming motion, giving the artifact its name. Because of the dependence on the array azimuth, it is easy to understand why the swimming artifact can be reduced by combining multiple array images.

Depth phases can cause additional artifacts in the form of "ghost" emitters corresponding approximately to the bounce points of the surface-reflected phase, but they also can carry additional information. For large intermediate-depth earthquakes, the time-delay between P and depth phases (e.g., pP and sP) allowed to improve the resolution in depth by combining both P and depth phase backprojections (e.g., Kiser et al., 2011). For more shallow earthquakes (40–100 km), depth phases can contribute significantly to uncertainties (e.g., Zeng et al., 2020). To our knowledge, however, a systematic imaging method that integrates depth phases and multiple arrays for the shallow depth range has not been reported yet.

In this study, we extended the multi-array approach of Rössler et al. (2010) by including depth phases depending on the radiation pattern (for earthquakes deeper than 40 km) and weighted seismic array images. We provide an

algorithm for automatically estimating rupture length, directivity, speed, and aspect ratio from backprojections. Short-period rupture lengths are additionally used to calculate scaling relations.

This modified method is applied to derive a catalog of rupture histories of recent large earthquakes (01/2010–12/2022) in the 0.5–2.0 Hz frequency range, which is complete for $M_W \ge 7.5$ and depths less than 200 km. The analysis focused on complex ruptures, depth-varying short-period radiation for large subduction earthquakes, and the detection of supershear ruptures for strike-slip events. We also showed that short-period ruptures encircling asperities are frequent in subduction megathrust earthquakes. The results suggest that encircling rupture patterns around slip patches are primarily related to the high-stress gradient around the asperity (and seismogenic barriers).

2. Methods

2.1. Multi-Array Multi-Phase Back-Projection

The back-projection method is similar to beamforming in maximizing the coherency of time-shifted waveforms at an array. Unlike in beamforming, there is no assumption of a planar wavefield. Instead, the time shifts are calculated from the predicted travel times for a grid spaced around the hypocenter. In practice, the grid is usually two-dimensional, chosen to be either a horizontal plane at the hypocentral depth, or a plane aligned with one of the nodal planes of the focal mechanism, or an a priori known fault surface, for example, the slab interface in subduction zones. The waveforms are back-projected onto this grid. The theoretical arrival of a target seismic wave (e.g., P wave) based on a reference velocity model controls the beamforming delays. For each grid point, the resulting array beam is:

$$b_i(t) = \frac{1}{N} \sum_{k=1}^{N} u_k(t + t_{ik} + \Delta t_k),$$
(1)

where *N* is the number of stations, $b_i(t)$ is the beam for the *i*th grid point, u_k the vertical component waveform recorded at station *k*, t_{ik} the travel time between the grid-point *i* and station *k* in a reference velocity model, and Δt_k the station-specific static correction term accounting for differences between the reference and true velocity model.

The station correction terms can, to a large extent, absorb the effect of 3D Earth heterogeneities on arrival times. They are usually determined by cross-correlating the first few seconds of the rupture recorded by each receiver. The resulting time-shifted arrivals are then compared to those predicted for the catalog hypocenter, and the differences correspond to the necessary correction terms. Thus, the back-projection image retrieves the rupture nucleation at the catalog hypocenter by definition. For very large earthquakes, aftershocks can alternatively be used to correct source-receiver paths away from the hypocenter (e.g., Ishii et al., 2007; Meng et al., 2016; Palo et al., 2014). The advantage over the hypocenter-based calibration is that the location errors of several events are averaged, such that a possible bias from mislocation of the mainshock hypocenter is reduced, and furthermore, spatially varying station terms due to 3D structure effects can be accommodated. The calibration with aftershocks is particularly important when the rupture pattern is compared with aftershocks. In contrast, the mainshock calibration offers advantages in near-real-time applications since only the hypocenter is required. Alternatively, travel time calculations could be performed in a 3D Earth model (e.g., Liu et al., 2017). For simplicity, in this work, we adopt the hypocenter calibration method.

In order to carry out the actual rupture tracking, the maxima of beamformed energy (\mathbf{E}_i) and semblance (\mathbf{S}_i), defined in Equations 2 and 3 below, are used to locate the most intense emission at each time step. The energy represents the amount of radiation emitted, and semblance provides a measure of the coherence of waveforms which is not affected by the amplitudes of individual traces and is, therefore, more effective for tracking the location of the earthquake rupture (e.g., Neidell & Taner, 1971; Palo et al., 2014; Rössler et al., 2010). For both measures, a time window of length *W* needs to be defined, which should accommodate at least two periods of the longest period analyzed.

$$\mathbf{E}_{\mathbf{i}}(t) = \int_{t}^{t+W} |\mathbf{b}_{\mathbf{i}}(\tau)|^{2} d\tau$$
(2)



$$\mathbf{S}_{\mathbf{i}}(t) = \frac{1}{N} \frac{\mathbf{E}_{\mathbf{i}}(t)}{\int_{t}^{t+W} \left[\sum_{k=1}^{N_{k}} \mathbf{u}_{\mathbf{k}}^{2}(\tau + t_{ik} + \Delta t_{k}) \right] d\tau}$$
(3)

Both measures provide an image of the earthquake rupture. Given a grid of sources, tracking the local semblance maxima provides a proxy of the rupture propagation. The energy peak in each time window provides a relative measure of the source time function of short-period seismic energy, which is related to but not necessarily proportional to the moment rate, for example, as derived from finite fault solutions. Thus, the time-integrated energy maps provide a summary view of the high-frequency energy radiation.

For the processing of data from multiple arrays, we initially follow the approach of Rössler et al. (2010) where semblance and energy maps from N_a arrays are multiplied to provide the rupture and energy radiated, respectively:

$$\hat{\mathbf{E}}_{\mathbf{i}}(t) = \prod_{a=1}^{N_a} \mathbf{E}_{\mathbf{i}}(t)$$
(4)

$$\hat{\mathbf{S}}_{\mathbf{i}}(t) = \prod_{a=1}^{N_a} \mathbf{S}_{\mathbf{i}}(t)$$
(5)

Here, we introduce two modifications: (a) we use exponents (equivalent to weights in log-space) to balance the contributions of individual arrays (based on their azimuthal distribution to avoid artifacts due to clustering of arrays in certain azimuthal ranges), and (b) for earthquakes deeper than 40 km, we combine P and pP back-projections to reduce artifacts related to the depth phase (see Figure 1a). The modified expressions are:

$$\hat{\mathbf{E}}_{\mathbf{i}}(t) = \prod_{a=1}^{N_a} \left[\prod_{p=1}^{N_p} f(t-t_0) \cdot \mathbf{E}_{\mathbf{i}}(t) \right]^{\gamma_a}$$
(6)

$$\hat{\mathbf{S}}_{\mathbf{i}}(t) = \prod_{a=1}^{N_a} \left[\prod_{p=1}^{N_p} f(t-t_0) \cdot \mathbf{S}_{\mathbf{i}}(t) \right]^{\gamma_a}$$
(7)

We assign p = 1 to P and p = 2 to pP waveforms, and N_p is the number of phases used. Here $N_p = 1$ is used for shallow earthquakes and 2 for deep earthquakes, but the method is open to experimenting with alternate seismic wave arrivals (e.g.,: sP, PP, and PKIKP; e.g., Kiser et al., 2011; Koper et al., 2012). The weighting exponent γ is set proportionally to the sum of the two half-angles between the azimuths of target and neighboring arrays (see Figure 1b), where the median of the array receiver coordinates is assumed to be the reference location for the weighting, and the normalization is chosen such that $\sum_{i}^{N_a} \gamma_i = 1$. Therefore, for a single array $\gamma = 1$, and for two, we always have $\gamma = 0.5$, but for irregularly shaped arrays, the weighting depends on the distribution. This weighting ensures internal consistency in the sense that for a perfectly coherent signal, a large array could be split into two sub-arrays, and the resulting peak energy values would be the same as would have resulted from a single array. The same principle can be extended to combine backprojections from arrays symmetrically well-located around the source. However, the resulting source image might be biased due to a non-symmetric configuration, that is, clustering of arrays in a specific azimuthal range. Here, a weighting scheme based on the azimuthal coverage allows for balancing the contribution of each array, preventing potential artifacts from affecting the solution. The term $f(t - t_0)$, with t_0 the origin time of the earthquake, is a taper function adapted from Kiser et al. (2011) to mute the waveform before the first arrival when multiple seismic phases are included:

$$f(\tau) = \begin{cases} 0 & \text{for } \tau \le -T/2 \\ \frac{1}{2} \left[\cos\left(\frac{2\tau\pi}{T}\right) + 1 \right] & \text{for } -T/2 < \tau < 0 \\ 1 & \text{for } \tau \ge 0 \end{cases}$$
(8)





Figure 1. Multi-array multi-phase back-projection. (a) Workflow for rupture imaging using P and pP waveform arrivals and multiple seismic arrays. pP backprojections are only included for earthquakes with depth \geq 40 km and are weighted according to radiation pattern (see text). The last step involves the extraction of a few rupture parameters based on the timing and locations of rupture maxima. (b) Array weighting. The sum of azimuthal half-angles between the target and its two neighboring arrays is proportional to the weights γ . (c) Automatic estimation of the rupture aspect ratio, speed, length, and directivity; see text for details.

If $\tau = 0$, the rupture has initiated, and the results are fully incorporated to image the source; otherwise, they are suppressed. *T* is the period of the cosine taper function controlling the transition between admitted and suppressed intervals.

2.2. Processing Details

Fifty-six large earthquakes in the time range 01/2010–12/2022 (all events with $M_W \ge 7.5$, depth ≤ 200 km according to Global CMT catalog, https://www.globalcmt.org) were back-projected. Figure 2 shows the distribution of the earthquakes color-coded by faulting type. Thrust earthquakes dominate the catalog (57%), followed by strike-slip (27%) and normal (16%) faulting mechanisms. Tables S1–S3 in Supporting Information S1 present the earthquake source parameters for strike-slip, thrust, and normal faulting earthquakes, respectively.

The number of (ad-hoc) arrays weighted and combined in the analysis depended on their availability at epicentral distances between 30 and 100°, where the P waveforms are not subject to complications like triplications as observed at smaller distances. The array selection and weighting prioritize azimuthal distribution while incorporating as many arrays as possible.





Figure 2. Global distribution of back-projected earthquakes colored by focal mechanism. Earthquakes back-projected with the combined P and pP approach are highlighted in magenta color. Labels designate the earthquake ID, with details given in Figure 4 and Tables S1–S3 in Supporting Information S1. Red segments show major plate boundaries (Bird, 2003).

Dense arrays were formed depending on the configuration and geometry of permanent broadband networks, for example, North America (e.g., US: United States National Seismic Network; AK: Alaska Regional Network), Japan (NIED Hi-net Network), Europe (many national and global networks distributed by ORFEUS-EIDA; Strollo et al., 2021) and Africa (e.g., AF: AfricaArray). Additionally, local and regional networks (often temporary deployments) were evaluated (e.g., data availability, waveform coherence) to form smaller aperture arrays and maximize the azimuthal coverage. Therefore, the extent to which a good azimuthal distribution can be achieved differs between regions and also depends on the time and time gap after the event (as temporary network data are often only openly available a few years after the experiment). Earthquakes in Indonesia were favorably located to be within range of several arrays, for example, networks in Asia, Europe, Africa, Australia, and Antarctica. In contrast, earthquakes in northern and central Chile suffered from limited coverage, although a somewhat reasonable azimuthal distribution can be achieved by combining networks in North America, Africa, and Antarctica. For events in southernmost Chile, Australian networks improved the coverage.

The data pre-processing involves mainly the removal of the instrument response from the waveforms and basic quality assessment. The mainshock location, origin time, and theoretical P and pP wave arrival times controlled the rupture imaging, allowing a reference frame to merge multiple array backprojections. P and pP wave arrival times were predicted based on the IASP91 velocity model (Kennett & Engdahl, 1991). For P waves, static corrections were determined by measuring the relative time shifts of first arrivals with the adaptive stacking method of Rawlinson and Kennett (2004). We considered the first 15 s after the P-wave onset on bandpass-filtered (0.4–3.0 Hz) vertical velocity waveforms. This frequency band is a little wider than the band used for the back-projection (0.5–2.0 Hz) and optimizes coherence while enough high frequencies are retained for a precise alignment. For pP waves, theoretical arrival times were calibrated with the P wave corrections. We also removed anomalous traces that could impact the waveform stack during the adaptive stacking as part of the input quality control.

We combined P and pP backprojections for earthquakes deeper than 40 km and only if the theoretical pP wave amplitude reached a minimum of 40% of the direct P wave amplitude; see Section 3.2 for synthetic tests and extended discussion on the radiation pattern. This provides a minimum separation time between P and pP wave arrivals of ~11 s (depth \geq 40 km; arrays at epicentral distances of 30–100°) in the IASP91 Model (Kennett & Engdahl, 1991). The \geq 40 km depth threshold is additionally aimed to mitigate the impact of depth phases in the back-projection, as observed for intermediate-depth (40–100 km) earthquakes (e.g., Zeng et al., 2020). To calculate the pP/P amplitude ratio, we considered the radiation pattern for P and pP waves (based on the respective Global CMT focal mechanism), including the P-P wave reflection coefficient of the free surface derived from the Aki and Richards approximation (Aki & Richards, 2002). The P wave slowness was calculate the surface reflection To calculate the al., 1999). To determine the take-off angles at the source and calculate the surface reflection

coefficient, we used averages from upper, middle, and lower crustal layers of the CRUST2.0 model (Vp = 6.4 km/s, vs. = 3.5 km/s, and Density = 2.9 g/cm³; Laske et al., 2013). We considered the peak amplitudes evaluated over the array to calculate the pP/P amplitude ratio. Figure S9 in Supporting Information S1 shows the P-P wave reflection coefficient as a function of the incidence angle. Table S7 in Supporting Information S1 presents the pP/P amplitude ratios for the events and arrays included in the combined P and pP back-projection.

For strike-slip, normal, and thrust (non-megathrust) earthquakes, the target grid was placed on a horizontal plane at a reasonable depth to minimize artifacts, that is, the hypocentral depth, with grid points every 5 km. For megathrust earthquakes, the grid followed depth variations of the SLAB1.0 model (Hayes et al., 2012). We remind that depth is not resolved explicitly, but the horizontal position is translated to depth to obtain accurate theoretical travel time predictions. For bilateral ruptures of a large extent (>400 km), we additionally mapped semblance maxima over a pre-defined sub-region of the grid to probe secondary rupture patterns, that is, the 2010 Maule (M_W 8.8), 2011 Tohoku-Oki (M_W 9.1), and 2017 Komandorsky Islands (M_W 7.8) earthquakes. We also added the 2015 Illapel (M_W 8.3) earthquake and defined sub-regions along-dip to simultaneously track updip and downdip rupture emissions as reported by previous back-projection studies, that is, Meng et al. (2018).

In our back-projection analysis, we considered a moving window of length W = 6 s, chosen to be three times the dominant period (back-projection frequency band: 0.5–2.0 Hz) and moving forward in 1 s increments. Longer values for W would have over-smoothed the rupture image, and shorter ones result in instability. The time reference is chosen such that the time axis refers to the beginning of the sliding window used in the source region with zero sets to the nominal origin time of the event. This choice of reference means a coherent arrival will become visible at nominally negative times. Prior to back-projection, we applied a 0.5–2.0 Hz bandpass filter, corresponding to the highest frequency range for which consistently sufficient waveform coherency has been obtained in previous rupture imaging studies (e.g., Meng et al., 2015, 2016, 2018; Palo et al., 2014).

Finally, the end of the rupture was visually determined, but an automatic estimation was included for comparison. Reactivation of earlier fault segments or scattered semblance maxima were considered to indicate the end of the rupture. The apparent source time function of radiated energy was also considered in the visual inspection, as small values compared to its peak indicate the end of the rupture. For the automatic rupture end determination, we followed the approach of Xie and Meng (2020), which also relies on detecting scattered maxima. The earthquake rupture is declared to have ended when the standard deviation of the rupture emission points within a 30 s window exceeds a threshold of 50 km. Additionally, we defined three criteria to avoid under- and overestimated rupture end times. First, we included lower and upper boundaries to restrict the search window after the earthquake initiation relative to magnitude and faulting type. Specifically, we assume a linear rupture propagation and estimate the rupture length following the scaling relations of Blaser et al. (2010). We, therefore, convert the upper and lower bounds of rupture speed (6.0 and 1.5 km/s, respectively) into upper and lower bounds for rupture durations. Second, if the threshold in the variance can not be passed, it will be reduced by 1 km until a peak value can be reached. This could be necessary for events with compact rupture lengths that would not allow passing a significant threshold in the spatial variance. Third, complex events, for example, bilateral ruptures or complex fault systems, might present high variance in the location of rupture emissions, which would wrongly indicate the end of the rupture. After inspecting the processed events, we ignore early variance peaks up to a specified time delay, specific to each event.

2.3. Estimation of Basic Source Parameters

We provide automated estimates of rupture length, directivity, and speed based on the rupture propagation derived from semblance maxima. Furthermore, we calculate aspect ratios to provide a proxy of rupture complexity. We emphasize that these parameters are only reflective of the high-frequency behavior of the earthquake and cannot always be assumed to coincide with the long-period properties based on total slip. Nevertheless, at least for length and directivity estimates, we expect a reasonable correlation (see discussion in Section 3.5.3). Clearly, if spurious scattered semblance maxima are included, errors in these estimates can become quite large, and it is, therefore, important to carefully determine the end of the rupture as accurately as possible. In our analysis, we used the manually determined rupture end times. Results for automatically determined rupture terminations are presented in Supporting Information S1.

To quantify rupture length (see Figure 1c), candidate lengths L are estimated by the projection of the semblance maxima (blue-red circles in Figure 1c) on lines passing through the epicenter (white star) for all azimuths $0-180^{\circ}$

(magenta line and open black circles show the realization for a 70° azimuth). The maximum value of *L* over all azimuths is chosen as the rupture length (red line in the bottom left subplot in Figure 1c).

The aspect ratio measures the sparsity of the short-period rupture emissions and is defined by the quotient between the minimum and maximum length estimates obtained from the rupture length grid search (aspect ratio close to 1: significant sparsity; aspect ratio close to 0: very low sparsity). We emphasize that the aspect ratio does not attempt to describe the earthquake dimension or rupture area, which is not well represented by high-frequency emissions, but the complexity, especially for larger earthquakes ($M \ge \sim 8$). Specifically, a large aspect ratio indicates a high degree of sparsity, for example, a conjugate fault rupture as observed for some oceanic intra-plate earthquakes (e.g., 2012 Wharton Basin and 2018 Gulf of Alaska) or rupture emissions outlining a simple megathrust earthquake asperity, that is, encircling rupture patterns. Subduction earthquakes tend to have shallowly dipping rupture planes, such that their surface projection represents approximately the fault plane. Here, the aspect ratio represents the actual aspect ratio of the high-frequency rupture and is sensitive to whether emissions occur, for example, downdip, updip, or encircling the main slip patch. Due to their much shorter rupture lengths and thus higher relative sensitivity to scattered emissions, the aspect ratio of events smaller than $M \sim 8$ can not usually be reliably estimated, but for consistency, it is reported for all analyzed earthquakes.

Directivity corresponds to the average direction in which the rupture fronts propagate. It is measured similarly to rupture length, with lines of all azimuths 0–180° pivoting through the epicenter, but the quantity minimized here is the sum of the squares of the perpendicular distances of all semblance maxima to the line (i.e., average squared lengths of blue arrows in Figure 1c, which are shown symbolically for a few semblance peaks only). For simple ruptures, the azimuths returned by the length and directivity measurements will be very similar, but the length estimate is controlled by the end points, whereas the directivity estimate is controlled by all points simultaneously. The ambiguity in the actual directivity is resolved by considering an imaginary line perpendicular to the rupture direction and centered on the epicenter; we then choose the directivity based on which side of this line more rupture emission points are found.

To determine the rupture speed, we first calculate estimates of instantaneous rupture velocities by dividing the distance between the epicenter and a subsequent semblance peak by the time elapsed since the origin time, which really represents the average rupture velocity from nucleation to the current time step. The resulting time series is strongly affected by location uncertainties for small distances and times but stabilizes quickly (see Figure 1c). The time series of rupture velocities is smoothed within a 4 s wide time window, and its peak value is then taken as the event rupture speed; maxima during the first unstable 10 s are ignored. Although strictly speaking, this estimate physically represents the maximum average velocity (where the average is taken over all preceding times) and represents a reasonable estimate for most ruptures. Based on theoretical considerations, for supershear ruptures on simple faults, supershear velocities are thought to be attained quickly and maintained until the end of the rupture (Das & Aki, 1977), such that they can be detected straightforwardly by this estimation procedure. Nevertheless, the estimated rupture velocities can be significantly lower than the peak rupture speeds, particularly for complex ruptures with directional changes or those with a slow start. To mitigate the effect of a slow start caused by a small triggering nucleation event, for example, the 17 December 2016 Solomon Islands (M_W 7.9) earthquake, instead of referring back to the epicenter for the whole rupture duration, we reset the reference point to an emission point at the beginning of the main phase of the rupture.

We also calculated average rupture velocities for comparison with our preferred maximum values. While average velocities can serve as a reference for assessing the typical estimated velocity in each earthquake rupture, the use of maximum velocity estimates enables the evaluation of rupture velocity transitions, such as those from subshear to supershear rupture. These transitions may be suppressed or underestimated when relying on averages.

3. Results and Discussion

3.1. Case Study: The 2020 Kuril Islands Earthquake

We introduce the presentation of our results based on the 25 March 2020 East of Kuril Islands (M_W 7.5) earthquake as an example (Figure 3). Results for the complete catalog are presented in Figures S12–S67 of Supporting Information S1, including a summary of derived parameters in Figure 4 and Table S4 in Supporting Information S1. We selected the Kuril Islands event because of the relatively simple rupture and because, at the time of writing, to our knowledge, no other back-projection analysis had been published yet for this event. This event is an





Figure 3. The 25 March 2020 East of Kuril Islands (M_W 7.5) earthquake back-projection (0.5–2.0 Hz). ID: 16—Origin Time: 2020-03-25 02:49:21.2 UTC—Lon: 157.70°E—Lat: 48.96°N—Depth: 58 km—Type: thrust. (a) Rupture pattern. Blue-red dots show semblance maxima tracking the earthquake rupture color-coded by time and scaled by energy radiated. Black dotted contours outline the short-period energy radiated (normalized to 1). The yellow-red polygons present the USGS-NEIC finite fault slip solution for comparison. Trench line derived from SLAB1.0 model (Hayes et al., 2012). Focal mechanism from Global CMT catalog. Inset: Array weights and energy radiated source time function. (b) Multi-array configuration and time-shifted P and pP waveforms. (c) Automatic rupture parameter determination; see Figure 1c for further information on the format.

intraplate thrust earthquake, which probably was triggered by compressional bending stresses in the deep interior of the subducting Pacific plate (Ye et al., 2021).

The left inset of Figure 3a shows the distribution of arrays, where the arrays in North America and Australia are weighted more strongly as they cover a larger backazimuthal range. In Figure 3b, the waveform coherency near the rupture initiation is visualized. It is generally good except for the pP phase at the China array. The main plot compares the back-projected rupture with the USGS-NEIC finite fault slip solution (https://earthquake.usgs.gov/earthquakes/eventpage/us70008fi4/finite-fault); for other events, frequently finite slip models from the literature are shown instead. Because this event is not a subduction megathrust earthquake, we back-projected onto a horizontal plane at 58 km depth, the hypocentral depth of the event.

For the East of Kuril Islands earthquake, the finite fault solution of Ye et al. (2021) favored rupture on the northwest-dipping fault plane. A further advantage of a horizontal plane is that this assumption will not fail in case of more than one planar fault being activated (e.g., the 2018 Gulf of Alaska earthquake; Figure S52 in Supporting Information S1). For the Kuril event, the main slip patch appeared close to the epicenter, and at the beginning of the rupture, both finite slip and short-period rupture spatially agreed. After 15 s, the high-frequency rupture propagated to the east of the secondary slip patch. This time period revealed the most energetic short-period emission. Close to this area, the most intense aftershock activity was observed (see Figure 1 of Ye et al. (2021)).





Journal of Geophysical Research: Solid Earth

Aspect HE Bupture Maximum

	ID	Time (UTC)	Mw	Length	Ratio	Directivity	Speed	%Vs	
CHIAPAS NORTHERN PERU RAT ISLANDS KHASH PAPUA NEW GUINEA OKERMADEC ISLANDS N HONSHU (OUTER RISE) ECUADOR-PERU BORDER LOYALTY ISLANDS	56 55 54 53 52 51 50 49 48	2017/09/08 04:49 2019/05/26 07:41 2014/06/23 20:53 2013/04/16 10:44 2022/09/10 23:46 2011/07/06 19:03 2011/03/11 06:25 2019/02/22 10:17 2018/12/05 04:18	8.2 8.0 7.9 7.6 7.6 7.6 7.5 7.5			334° 345° 314° 299° 1° 8° 216° 310°		80% 82% 44% 71% 49% 51% 57% 38% 68%	Supershear
TOHOKU-OKI MAULE SOUTH OF SANDWICH I. ILLAPEL CHIGNIK KERMADEC ISLANDS IQUIQUE-PISAGUA SOLOMON ISLANDS SANTA CRUZ ISLANDS GORKHA IBARAKI-OKI SOLOMON ISLANDS SANTA CRUZ ISLANDS GORKHA IBARAKI-OKI SOLOMON ISLANDS NORTHERN SUMATRA MENTAWAI HAIDA GWAII SIMEONOF ISLANDS IQUIQUE PHILIPPINE ISLANDS NICOYA MELINKA OMETEPEC SOUTH GEORGIA ISLAND PAPUA NEW GUINEA NEW BRITAIN REGION NEW BRITAIN REGION SOLOMON ISLANDS KURIL ISLANDS	47464444444444444444444444444444444444	2011/03/11 05:46 2010/02/27 06:34 2021/08/12 18:32 2015/09/16 22:54 2021/07/29 06:15 2021/03/04 19:28 2014/04/01 23:46 2017/01/22 04:30 2016/12/17 10:51 2013/02/06 01:12 2015/04/25 06:11 2011/03/11 06:15 2016/12/08 17:38 2010/04/06 22:15 2010/10/25 14:42 2016/11/13 11:02 2016/04/16 23:58 2012/10/28 03:04 2020/07/22 06:12 2022/09/19 18:05 2021/02/10 13:19 2014/04/03 02:43 2012/08/31 12:47 2012/09/05 14:42 2016/12/25 14:22 2012/03/20 18:02 2016/08/19 07:32 2018/02/25 17:44 2015/03/29 23:48 2014/04/19 13:28 2020/03/25 02:49	9.8.8.3.2 8.8.8.2.1 9.9.9.9.9.9.8.8.8.8.7.7.7.6.6.6.5.5.5.5.5.5.5.5.5.5.5.5.5.5	110000000000000000000000000000000000000		228° 29° 195° 11° 16° 131° 145° 143° 143° 281° 341° 210° 317° 39° 191° 264° 301° 264° 301° 264° 301° 28° 357° 29° 240° 341° 416° 293° 41° 41°		$\begin{array}{l} 54\% \\ 577\% \\ 577\% \\ 672\% \\ 851$	O B5-105%Vs: Likely O Hypocentral Depth (km) Vs speed (km/s)
WHARTON BASIN WHARTON BASIN GULF OF ALASKA KOMANDORSKY ISLANDS* SCOTIA SEA BALOCHISTAN WHARTON BASIN WHARTON BASIN WHARTON BASIN WHARTON BASIN SOLOMON ISLANDS SAND POINT PALU NICOBAR ISLANDS NORTH OF HONDURAS	15 13 12 10 9 8 7 6 5 4 3 2 1	2012/04/11 08:38 2012/04/11 10:43 2018/01/23 09:31 2017/07/17 23:34 2013/11/17 09:24 2013/09/24 11:29 2020/01/28 19:10 2019/05/14 12:58 2014/04/12 20:14 2020/10/19 20:54 2018/09/28 10:02 2010/06/12 19:26 2018/01/10 02:51 2013/01/05 08:58	8.6 8.2 7.8 7.8 7.7 7.6 7.5 7.5 7.5 7.5 7.5	0.000000000000000000000000000000000000		261° 214° 95° 224° 224° 133° 132° 287° 171° 287° 171° 233° 2339°	A A A A A A A A A A A A A A A A A A A	89% 67% 62% 86% 74% 98% 43% 104% 53% 37% 43% 125% 57% 95% 137%	0 Rupture Time (s)

Figure 4. Earthquake rupture parameters from back-projection (0.5–2.0 Hz). Event ID (for cross-referencing to other figures, tables, and Supporting Information S1), moment magnitude, rupture length (colored by rupture time), aspect ratio, directivity, and rupture speed (colored by hypocentral depth). The events are categorized into normal (upper), thrust (middle), and strike-slip (bottom) earthquakes, with the order in each category determined by magnitude. Events in bold text are assumed to be subduction megathrust earthquakes. Events labeled with a black bullet indicate that a secondary rupture pattern was obtained over a predefined grid and included in the rupture parameter estimates. The vertical red bars in the rupture speed panel show the expected shear wave speed (V_S) at the hypocentral depth in the IASP91 velocity model (Kennett & Engdahl, 1991). Rupture speeds in the range 95–105% V_S and >105% V_S define "likely" and "strongly" supershear earthquakes, respectively.

The rupture propagated unilaterally along a nearly straight line, making the length and directivity estimation straightforward (Figure 3c). The implied strike direction of the rupture track is clockwise rotated by some $10-20^{\circ}$ with respect to the strike direction indicated by the moment tensor. At face value, this would imply that the short-period emission points did not strictly propagate along strike but moved additionally updip (or downdip, depending on which fault plane is the correct one). The automatically determined rupture duration was 2 s longer than the visually determined end time (see Figure S1 in Supporting Information S1 for automatic end times).





Figure 5. Short-period (0.5–2.0 Hz) energy radiated source time functions derived from back-projection. The energy radiated maxima evaluated over the rupture time provides the source time function (in gray). Additionally, SCARDEC (in blue) and USGS-NEIC (in red) moment rate functions are shown for comparison. The event highlighted with a dashed box is used as an example and discussed in detail in the text.

Finally, we can compare the time history of emitted short-period energy with the moment rate obtained from the SCARDEC and USGS databases (highlighted in Figure 5). The functions show two peaks, and the timing of the first peak and subsequent trough agree. However, at short periods, the second energy peak exceeds the first one, unlike for the USGS and SCARDEC solutions, and the rupture appeared to continue for longer. Interestingly, in the source time function estimated by Ye et al. (2021) (their Figure 1), the moment rate of the second peak also exceeded the first one.

3.2. Synthetic Tests

To evaluate the lateral resolution, we considered the array configuration used in the Kuril Islands earthquake back-projection (see Figure 3b) and simulated a unilateral rupture scenario. Synthetic seismograms were generated using a Ricker wavelet with a central frequency of 1 Hz and 50% of Gaussian random noise (e.g., Kiser et al., 2011; Ricker, 1953). The maximum amplitude of the signature defined the standard deviation of the Gaussian noise. Five sources placed every 15 km at the hypocentral depth of the Kuril event were used to simulate





Figure 6. Back-projection of northeast-propagating rupture synthetics with variations in the number of combined arrays. (a) Synthetic rupture scenario. (b) Synthetic waveforms for the European (EU), North American (USA), Australian (AU), and China (CN) array. A pP/P amplitude ratio of 1 is assumed. (c) P wave (top panel) and combined P and pP (bottom panel) backprojections. The number of combined arrays increases from left to right. The white stars mark the location of the emission points. Inset: Array weights and energy radiated source time function. Rupture parameters estimates; rupture speed, length, directivity, and aspect ratio.

the rupture propagation. Although using only five separated sub-sources does not lead to realistic-looking seismograms, this scenario allows us to visualize the accuracy of the recovered rupture easily.

To observe the effect of merging arrays, we assumed P and pP waves with the same amplitude and simulated a rupture propagating to the northeast at 2.5 km/s. Figure 6 presents tests using only P (upper subplots) and combined P and pP (lower subplots) backprojections with a variable number of contributing arrays. The resolution improves notably with the number and wide azimuthal distribution of combined arrays. Furthermore, the combined P and pP back-projection can better resolve the rupture compared to the P-only back-projection, also resulting in more accurate rupture length and directivity estimates. Rupture speeds are often mildly overestimated with variations up to 22% with respect to the true speed (pP-P back-projection; Figure 6). We remind that the rupture speed corresponds to the maximum average velocity evaluated over the rupture time. Overestimated rupture speeds, that is, the maximum average speed, may arise or be affected by artifacts or smearing effects between the activation of the source points and the initial unstable first few seconds of the rupture tracking. We preferred maximum over average speeds to quantify the uppermost speed limit from short-period back-projection. This also allows us to identify supershear ruptures that could be underestimated with average values while providing a standard frame to evaluate the earthquake database. It should also be noted that in the tests of Figure 6 and Figures S4-S6 in Supporting Information S1 (described below), P and pP Ricker wavelets have the same amplitude. The variations in the amplitude of the waveforms (Figure 6b) correspond to interference between P and pP wavelets, which arises from the set of emission points used for simulating the rupture propagation.

We tested other scenarios with rupture propagating to the southwest (Figure S4 in Supporting Information S1) or bilaterally (Figure S5 in Supporting Information S1) and observed similar improvements. For the bilaterally propagating rupture, the addition of the AU and CN arrays continues to improve the estimate, while for the northeast propagating rupture, only marginal improvements were observed from adding the third and fourth arrays. For the northeast-propagating rupture scenario, we also tested the pP back-projection alone to assess the distorting effect of P waves from the later part of the rupture on the pP back-projection image (Figure S6 in Supporting Information S1). Tracking the rupture initiation is challenging for the pP back-projection but overall the result is comparable with the P back-projection by merging multiple arrays.

We also conducted synthetic experiments with different pP/P amplitude ratios, which in reality depend on the radiation pattern. We tested the northeast-propagating rupture scenario shown in Figure 6. If the pP wave amplitude is at least as large as that of the P wave, the combined P and pP approach performs better than only the P back-projection with regard to the accuracy of rupture length and directivity estimates and the tracking of emission points (see Figure 7). If the pP amplitude drops to 40% of the P wave amplitude, the resolution decreases, but consistency is still comparable to the input. When the pP amplitude ratio drops to 30% of the P wave amplitude, the combined P and pP back-projection fails to separate emission points properly, and the results of the P-only back-projection are preferable.

We also tested slower and faster rupture speed scenarios of 1.5 and 4.0 km/s. If the rupture propagates at 4.0 km/s, the advantages over the P back-projection remained, but both P and combined pP and P backprojections failed to recover the beginning and end of the rupture, thus underestimating its length (Figure S7 in Supporting Information S1; the input rupture length is underestimated by up to 26%). The variations in the recovered speed are 17% of the input speed. The back-projection for a slow rupture scenario of 1.5 km/s is more prone to artifacts (Figure S8 in Supporting Information S1). Scattered emissions points are frequent, and rupture speeds are severely overestimated (up to 71%).

Following the tests, as previously mentioned in the description of the processing workflow (Section 2.2), we exclusively applied the combined P and pP approach for events deeper than 40 km depth and only for those arrays where the predicted pP wave amplitude is at least 40% of the P wave; otherwise, the P-only back-projection is used. Additionally, the tests suggest resolving emission points ~15 km apart (horizontally) should be possible in most cases.

Furthermore, we tested the back-projection with a selection of aftershocks to provide a more realistic rupture simulation in terms of noise and complexity of the waveforms. We considered a sequence of eight aftershocks that approximately covered the 2011 Tohoku-Oki (M_W 9.1) earthquake extent. Please refer to Table S8 in Supporting Information S1 for details on the aftershock source parameters and Figures S68–S69 in Supporting Information S1 for their visualization (aftershocks marked with stars, outlined in cyan color). Before stacking, we pre-



Journal of Geophysical Research: Solid Earth



Figure 7. Back-projection of northeast-propagating rupture synthetics with variations in the pP/P amplitude ratio. P (top panel) and combined P and pP (bottom panel) backprojections. The pP/P amplitude ratio increases from left to right, expressed in percentages. Input rupture speed equivalent to 2.5 km/s. The white stars mark the true emission points. Inset: Array weights and energy radiated source time function. Rupture parameters estimates; rupture speed, length, directivity, and aspect ratio.

filtered the data within the 0.1–5.0 Hz frequency range for display purposes, which is a wider band than the one later implemented in the back-projection (0.5–2.0 Hz). We also normalized the waveforms to prevent an image dominated by the largest aftershocks, which could be larger than realistically expected for a single rupture. Thus, normalization allows for a balanced contribution of individual aftershocks and ensures that the back-projection tests focus on rupture tracking. Due to the significant distance between the emission points, ranging from 38 to 85 km, we conducted tests with propagation speeds of 2.5 and 4.0 km/s, that is, unilateral and bilateral rupture scenarios. This provides gaps up to ~90 s in the rupture propagation (2.5 km/s: gaps between 21 and 93 s; 4.0 km/s: 13–58 s), allowing to assess the performance of the back-projection when dealing with interruptions in rupture emissions, that is, potential smearing artifacts. Following the processing of rupture synthetics, we performed P and pP backprojections in the 0.5–2.0 Hz frequency range. We conducted P·pP backprojections without considering the threshold depth of \geq 40 km and the radiation pattern (aftershock depth ranging from 19 to 53 km; Table S8 in Supporting Information S1). This evaluation aimed to assess the performance of the method in challenging rupture tracking scenarios, complementing the previous Ricker wavelet analyses while keeping the same aftershocks and station coverage used for the P back-projection tests. Because of the temporal span of the

aftershocks (03/2011–06/2012; Table S8 in Supporting Information S1) and the teleseismic waveform coherence for moderate magnitude events (M6.1–7.1; Table S8 in Supporting Information S1), we obtained only three arrays, well distributed in azimuth, that recorded all of the aftershocks, that is, EU (58–86 stations), USA (41–89 stations) and IND (4 stations). See Figures S70–S78 in Supporting Information S1 for the waveforms and array distribution used in the tests. Additionally, Figures S71–S73 in Supporting Information S1 presents an example of the stepwise inclusion of aftershocks to generate the waveforms shown in Figure S70 of Supporting Information S1 and back-projected in the tests of Figures S68a–S69a in Supporting Information S1. This illustration highlights the impact of the coda waves in generating complex rupture synthetics through a sequence of aftershocks. The limited number of stations and the complexity of the waveforms provided an additional challenging input to be back-projected and recovered.

Figure S68 in Supporting Information S1 shows the P wave backprojections for the selection of aftershocks, that is, southwest (left column), northeast (central column), and bilateral (right column) propagating ruptures at 2.5 km/s (upper subplots: a, b, and c) and 4.0 km/s (lower subplots: d, e, and f). The array coverage and waveform coherence for each scenario are presented in Figures S70–S75 of Supporting Information S1. The difference in the recovered rupture length and directivity relative to the input was relatively small, with an absolute variation of up to 18% for rupture length and 4° for directivity (rupture parameter estimates inset in Figure S68 of Supporting Information S1). The recovered rupture speed is overestimated by up to 31% compared to the input speed (as seen in the time series inset in Figure S68 of Supporting Information S1). However, it is important to consider the effect of artifacts when interpreting peaks in rupture speed, specifically, the smearing of emissions when the rupture stops propagating between the source points. Overall, the rupture propagation can be recovered in both space and time, even in more complex bilateral rupture scenarios (Figures S68c and S68f in Supporting Information S1; see the comparison between the back-projected rupture and the input sources, color-coded relative to activation time). The results for combined P and pP backprojections are presented in Figure S69 of Supporting Information S1. Compared to the P back-projection, the main difference in the rupture parameters is an overestimation of speeds (up to 52% of the input speed). However, it is worth noting that most of the rupture speed time series are below the input speed (see magenta dashed line in the speed time series; Figures S68 and S69 in Supporting Information S1). Potential smearing effects due to gaps in the emission points and the observed tracking instability during the first few seconds of the rupture should be considered when interpreting rupture speeds and overall rupture parameter estimates.

The Ricker wavelet and aftershock sequence tests showed the reliability of combining different array and P and pP wave backprojections (depth \geq 40 km; pP wave amplitude \geq 40% of the P wave) for tracking the earthquake rupture. This is observed even in complex bilateral rupture scenarios and when arrays of limited aperture complement the processing.

3.3. Short-Period Source Time Functions

The short-period energy time functions for all events are compared with USGS and SCARDEC (Vallée & Doeut, 2016) moment rates in Figure 5. The time history of radiated high-frequency seismic energy frequently agrees (e.g., 2021 Chignik, ID 43; 2020 Caribbean, ID 8; 2016 Kaikoura, ID 32; 2013 Scotia Sea, ID 11; 2012 Wharton Basin, ID 15; 2010 Mentawai, ID 33) with the seismic moment rate functions, but it is sometimes apparently time-shifted, following (e.g., 2016 Melinka, ID 23; 2013 Craig, ID 1; 2011 Tohoku-Oki, ID 47) or, less often, preceding (e.g., 2016 Solomon, ID 35; 2012 Philippine Islands, ID 25) the seismic moment rate. A few earthquakes presented quite different shapes (e.g., 2020 Kuril Island, ID 16). However, it has to be noted that the USGS and SCARDEC moment rates also do not always agree. One striking example is the Iquique-Pisagua (M_W) 8.1) earthquake (ID 41; Figure 5), where the SCARDEC estimate suggests a much shorter source time function. For the Chiapas (M_W 8.2) earthquake (ID 56; Figure 5), SCARDEC (and the short-period energy function) completely lacks the secondary peak seen in the USGS estimate. Because of this variability, it is difficult to judge whether the differences between moment rate and short-period energy function are physical (i.e., related to the time-varying ratio of slip to high energy emissions), methodological artifacts, or a combination of both. A possible physical explanation is that the ratio of short-period to long-period spectral energy is known to vary significantly between earthquakes (or equivalently, the seismic energy to moment relation, which is related to the variability in stress drop). It seems likely that this variability can also extend to the case of different asperities within a given rupture, which would then cause different time histories of moment rate and short-period energy function.



Journal of Geophysical Research: Solid Earth



Figure 8. Back-projected earthquake rupture classification for major megathrust earthquakes (0.5–2.0 Hz). Colored dots show semblance maxima, scaled by energy radiated: blue for maxima downdip of the main asperity, red for those updip, and gray for emissions inside the asperity (see text for details of the definition of asperity). The dashed black line indicates the transect along with the fault strike and maximum slip (magenta cross mark) used to classify the rupture pattern relative to the asperity (black contour). The subplots are sorted by rupture duration. The dark-magenta arrows (lower-right corner) indicate the multi-array distribution relative to the epicenter (white star). Trench location from SLAB1.0 model (Hayes et al., 2012); focal mechanisms from Global CMT catalog. The yellow-red background shows the slip distribution of (a) linuma et al. (2012), (b) Moreno et al. (2012), (c) Tilmann et al. (2016), (d), (e), (g) and (h) USGS-NEIC finite fault solution, (f) and (k) Schurr et al. (2014), (i) Crowell and Melgar (2020), (j) Heidarzadeh et al. (2017), and (l) Moreno et al. (2018). Bottom inset: High-frequency rupture pattern classification. The dominant type is noted in the upper of each plot.



3.4. Back-Projection of Subduction Megathrust Earthquakes

3.4.1. Megathrust Rupture Patterns

We now consider the rupture patterns of subduction megathrust earthquakes, specifically the spatial relationship between the short-period emissions and the finite slip models characterizing the long-period behavior. Previous observations indicated a strong preference for short-period emissions downdip of the main slip asperities (e.g., Lay et al., 2012). Figure 8 presents the rupture patterns for the largest subduction megathrust events in our catalog.

We can distinguish a variety of earthquake rupture patterns referred to as downdip, updip, and balanced (segmented or encircling) determined by the relation between short-period radiation and the asperity (see Figure 8 for a graphical definition; the same events are presented in Figure S10 of Supporting Information S1 color-coded by rupture time). We note that "encircling rupture patterns" are overall a purely kinematic description to highlight high-frequency emissions around megathrust asperities and does not necessarily mean a bifurcation of the rupture but could be, for example, the result of an extending rupture crack. Here, we define, somewhat arbitrarily, the asperity as the area enclosed by the contour corresponding to 75% of the maximum slip and classify each emission point whether it is downdip, updip or within the main asperity. For the downdip and updip classification, the emissions have to be outside the asperity contour and downdip or updip of a transect through the peak slip in the direction of the fault strike (see schematic overview in the lower-right part of Figure 8; the yellow-red color represents the slip patch, the encircling black contour define the asperity, and the black dashed line provides the reference for the updip and downdip emissions relative to the strike). If more than 75% of the short-period emission points are located downdip or updip from the asperity, then the rupture is classified as downdip or updip; otherwise, it is defined to be balanced. For balanced rupture patterns and based on visual inspection, we classify a rupture as balanced-encircling for emission patterns clearly outlining the asperity with a semicontinuous ring, and as balanced-segmented otherwise, that is, where there are noticeable gaps in the pattern of emission points (see visual representation in Figure 8).

The 2010 Maule earthquake represents a classical downdip rupture with a bilateral propagation (\sim 29% of the emission points are southward of the epicenter; Figure 8b and Figure S12 in Supporting Information S1). The 2011 Tohoku-Oki earthquake ruptured bilaterally and mainly downdip (Figure 8a and Figure S16 in Supporting Information S1). However, emissions were also observed near the trench where a large shallow slip occurred (up to 50 m). Toward the end of the rupture (135–180 s; Figure S16a in Supporting Information S1), a subsidiary pattern with updip and downdip emissions encircled the second larger slip area off the coast of Fukushima Prefecture (see the southern 10 m contour in Figure 8a). The 2010 Mentawai earthquake propagated predominantly along the downdip edge of the main asperity and then ruptured updip near the trench region (Figure 8d and Figure S15 in Supporting Information S1).

Notably, the observations include several events with updip and balanced-encircling patterns (see graphical definition inset in Figure 8). For the 2016 Pedernales earthquake, the reverse of the conventional pattern occurred, first with short-period emissions updip of the main asperity, followed by a downdip propagation near the southern tip of the rupture (Figure 8j and Figure S41 in Supporting Information S1). The 2021 Kermadec Islands earthquake initially propagated to the northeast along with the main asperity, followed by a rupture moving updip in the final phase (Figure 8h and Figure S63 in Supporting Information S1). The 2014 Iquique-Pisagua earthquake presented a balanced-encircling rupture around the larger slip area (Figure 8f and Figure S41 in Supporting Information S1); a similar sequence defined the 2016 Melinka earthquake (Figure 8f and Figure S47 in Supporting Information S1). The 2021 Chignik earthquake (Figure 8g and Figure S64 in Supporting Information S1) ruptured eastward and encircled the slip area. More or less parallel updip and downdip emissions outlined the main slip asperity for the 2015 Illapel earthquake (Figure 8c and Figure S39 in Supporting Information S1), consistent with the back-projection study of Meng et al. (2018) for this event. The main rupture propagated northward along the downdip. A secondary front in the shallow part of the megathrust (less than ~15 km depth) followed the region parallel to the trench for over ~150 km.

Other events exhibited balanced-segmented ruptures. The 2014 Iquique earthquake (main M_W 7.7 aftershock) showed a unilateral northeast short-period rupture partitioned around the asperity but still outlining the maximum slip region (Figure 8k and Figure S32 in Supporting Information S1). Offshore of the Alaska Peninsula, the 2020 Simeonof Island earthquake first propagated downdip to the northwest and west (Figure 8i and Figure S60 in Supporting Information S1), but the last short-period emissions are observed east of the epicenter.





Figure 9. Along-dip horizontal distance distribution of short-period radiation (0.5–2.0 Hz) for large megathrust earthquakes. Blue and red dots present updip and downdip short-period emissions relative to the earthquake asperity, which is defined by the area where the slip exceeds 75% of the maximum slip. Gray circles show emissions inside the asperity. White stars mark the horizontal distance from the epicenters. Earthquake magnitude and Global CMT centroid depths are shown on the top; earthquake rupture pattern classification on the bottom. The events are ordered by centroid depth.

To evaluate the depth distribution of short-period emissions more systematically, we present the along-dip horizontal distance distribution of radiation for the megathrust earthquakes shown in Figure 8; see Figure 9. Emissions located inside the asperity represented a median of only ~5% of the total emissions, with some exceptions, for example, the 2014 Iquique-Pisagua earthquake with ~48% of the radiation emitted within the asperity. For most events, short-period radiation is observed dominantly downdip, but the large variation in rupture patterns also demonstrated that the classical pattern of dominant downdip high-frequency radiation has exceptions. Notable examples are dominant updip short-period ruptures, for example, the 2016 Pedernales and 2021 Kermadec Islands earthquakes, and balanced ruptures, for example, the 2015 Illapel, 2016 Melinka, and 2021 Chignik earthquakes. There was no apparent relation between the distribution of short-period radiation and the centroid depth.

In addition, we compared multi-array backprojections for major megathrust earthquakes with IRIS backprojections. Specifically, we considered IRIS solutions from the Global Seismographic Network (GSN; http:// ds.iris.edu/spud/backprojection), which provides the tracking of the earthquake rupture from a wide azimuthal coverage. Figure S79 in Supporting Information S1 presents the IRIS backprojections (blue-red squares) compared to multi-array back-projections (blue-red circles, color-coded by rupture time) for the events shown in Figure 8. In general, the rupture tracking is comparable in both solutions regarding timing and directivity. However, the IRIS solutions have shown a higher susceptibility to artifacts, which is evident in the broader extent and sparsity of the rupture emissions. Thus, the combination of multiple arrays, weighted according to their azimuthal distribution, presented greater control over artifacts and, as a result, over the rupture tracking and the image of potential complexities. In the analysis, it is also important to consider the lower frequency band used in the IRIS-GSN solutions (0.05–0.25 Hz) relative to the multi-array back-projection (0.5–2.0 Hz). This lower frequency band could imply more significant smearing effects or artifacts due to the lower frequency range and, therefore, lower resolvability. The impact of the lower resolution, particularly the presence of artifacts, is more evident when estimating rupture parameters from IRIS backprojections. In effect, the results show similar directivity relative to this study but significantly larger rupture lengths and highly overestimated speeds (for a detailed comparison, refer to the rupture parameter estimates in Table S9 of Supporting Information S1).

3.4.2. Complex Megathrust Ruptures and Depth-Varying Short-Period Radiation

Seismicity prior to a great earthquake is often demonstrated to be active on the rupture zone edge while a seismic quiescence dominates the region later experiencing the maximum slip according to the "Mogi donut" hypothesis (e.g., Kelleher & Savino, 1975; Mogi, 1969). According to this hypothesis, the spots at the asperity edge become seismically active as the stress on the asperity increases, outlining the potential rupture area (e.g., Kanamori, 1981). For example, Schurr et al. (2020) observed a Mogi donut in the years prior to the 2014 Iquique-Pisagua earthquake. Sippl et al. (2021) showed microseismicity forming three half-ellipses along the central Chile megathrust, with one of them encircling the 2015 Illapel earthquake area. In back-projection, short-period rupture emissions are related to rupture velocity variations, in turn related to stress variations (e.g., Marty et al., 2019). They are often seen to concentrate around the main slip asperity and thus reflect presumably the high-stress gradient around the asperity, equivalently to the Mogi donut pattern of seismicity.

Many megathrust earthquakes exhibited balanced rupture patterns, where the back-projection highlights propagation to shallower and deeper regions relative to the asperity (peak slip) (Figure 9). This pattern could also be related to the presence of seismogenic barriers under the assumption of a strong rupture deceleration or stopping phase (e.g., Madariaga, 1983). For example, the 2015 Illapel event showed a balanced-encircling rupture (Figure 8c and Figure S39 in Supporting Information S1); it was first observed by Meng et al. (2018) from backprojection and interpreted as a splitting of rupture fronts surrounding a large asperity or barrier. Recently, a geometrical kink in the slab interface and the continental Moho depth were considered to explain rupture emissions both updip and downdip of the slip asperities for the 2007 Tocopilla earthquake; see Vera et al. (2023). The 2010 Mentawai (Figure 8d and Figure S15 in Supporting Information S1) and 2011 Tohoku-Oki (Figure 8a and Figure S16 in Supporting Information S1) earthquakes were tsunamigenic with a rupture that propagated up to the trench (e.g., Hill et al., 2012; Iinuma et al., 2012; Lay et al., 2011). This is compatible with the backprojections, where rupture emissions were partly located in shallow regions below accretionary structures. The transition from the seismogenic to a shallow aseismic sliding region might explain a strong rupture velocity variation and the shallow source of short-period radiation. For the 2016 Pedernales event, Agurto-Detzel et al. (2019) proposed a barrier mechanically controlled by the subduction of a rough oceanic relief in the updip limit of the earthquake rupture. Similarly, the back-projection showed a rupture updip of the main slip patch (at 15–20 km depth; Figure S41 in Supporting Information S1) that agrees with high residual (> 1 km) bathymetry data (see Figure 9 of Agurto-Detzel et al. (2019)), indicating that the strong short-period earthquake radiation might be related to this barrier structure.

In summary, short-period emissions from the shallow part of the megathrust interface, updip of the main slip asperities, are more frequent than suggested by earlier results.

3.5. Earthquake Rupture Parameters

3.5.1. Impact of Automatic and Manual Rupture Duration Estimates

Automatic and visually determined rupture end times show some agreement (see inset in Figure S1 of Supporting Information S1), but there is substantial scatter among the shorter rupture times, for example, for manually picked rupture times less than 50 s, the average difference relative to automatic times is 20 s. Automatically determined rupture durations are, on average, longer than manually determined durations, but underestimations also occur. Thus, manually picked end times are sufficiently superior to prefer them even though they introduce subjectivity. Figure S2 and Table S5 in Supporting Information S1 summarize the rupture parameters derived from the



automatic end times. Figure S3 in Supporting Information S1 compares the rupture parameters from the manually picked and automated results.

3.5.2. Rupture Speed

Figure 4 shows the rupture speed estimations derived from back-projection, additionally expressed as a percentage of the shear velocity at the hypocentral depth in the IASP91 model. The underlying rupture speed time series are summarized in Figure S11 of Supporting Information S1. Because the rupture speed estimates are based on the horizontal projection, rupture velocity might be underestimated if rupturing updip or downdip dipping faults, for example, for a dip angle of 30°, this effect would amount to ~15%. However, this effect will only affect a small subset of intra-plate thrust or normal faults.

Thrust earthquakes averaged a rupture speed of 2.1 km/s and 56% of the shear wave speed, with a range between 25% and 86% of the shear wave speed, placing propagation speeds firmly in the expected sub-Rayleigh regime. The 2010 Mentawai, 2011 Tohoku-Oki, and 2021 South of Sandwich Islands earthquakes were slower than the average. The Mentawai and Tohoku-Oki events are characterized by large displacement at very shallow depths, implying a low normal stress environment and large tsunamis. The Sandwich Islands event also presented a shallower rupture for at least part of the rupture propagation. This observation agrees with the multiple independent technique results presented in Metz et al. (2022), that is, a shallow and slow rupture (1.5–2.1 km/s) that follows the varying strike direction along the trench, as part of a complex rupture also involving a faster propagation on deeper parts of the plate interface.

The 17 December 2016 Solomon Islands (M_W 7.9) earthquake is a special case (Figure 8e and Figure S45 in Supporting Information S1). A megathrust rupture released the bulk of the moment, but its hypocenter is at 105 km depth, and the first motion mechanism indicated downdip intraslab faulting, which is thought to have triggered the main megathrust event (e.g., Lay et al., 2017; S.-J. Lee et al., 2018). In the back-projection, this sequence was visible as a ~55 km gap between the first emissions and the dominant megathrust rupture pattern. The short time interval needed to cross the gap would imply an extremely fast rupture speed of 5.7 km/s, equivalent to ~71% of the P wave speed at the hypocentral depth. This indicates that, most likely, P waves from the intraslab sub-event triggered the megathrust rupture (note that the relative P wave speed is calculated in the reference model, and P wave speeds in the down-going oceanic crust would be slower; thus, higher percentages are possible). To explore rupture propagation in the main phase, we placed the reference point at the megathrust rupture initiation (Figure S46 in Supporting Information S1); then, a more typical rupture propagation speed of 2.3 km/s was obtained.

For normal-faulting earthquakes, the average was 2.5 km/s and 60% of the shear wave speed, ranging between 38% and 82% of the shear wave speed. The 2019 Northern Peru (106 km depth; Figure S58 in Supporting Information S1) earthquake showed the fastest rupture (3.7 km/s and $82\%V_s$), which is at the upper limit of the sub-Rayleigh speed. Although this intermediate-depth event was fast, the depth does not appear to affect rupture velocity systematically. For example, the 2014 Rat Islands event at nearly the same depth (109 km; Figure S35 in Supporting Information S1) and the 2019 Ecuador-Peru Border Region earthquake (at 146 km depth; Figure S56 in Supporting Information S1) ruptured slowly at 2.0 and 1.7 km/s. For the Rat Island event, previous back-projections and finite fault modeling showed a similar rupture velocity in the range of ~1.5–2.5 km/s (e.g., Twardzik & Ji, 2015; Ye et al., 2014). However, the lack of solid evidence for a shallow or steeply-dipping fault plane preference hampers the evaluation of the projection effects in the rupture velocity estimation.

Strike-slip earthquakes showed the greatest variability. The average rupture speed was 2.7 km/s and 78% of the shear wave speed, but the range spans from 37% to 137%. This is comparable with the observations of D. Wang et al. (2016), where backprojections showed strike-slip earthquakes propagating at average speeds greater than 3.0 km/s, faster than dip-slip events. Two events were clearly within the unstable supershear range (between V_s and $\sqrt{2}V_s$): the 2013 Craig (4.6 km/s and 137% V_s ; Figure S26 in Supporting Information S1) and 2018 Palu (4.2 km/s and 125% V_s ; Figure S54 in Supporting Information S1) earthquakes. The Palu event was supershear from early on as reported by Bao et al. (2019) using the high-resolution MUSIC back-projection, while the oceanic interplate Craig initiated at subshear speed and transitioned to supershear after ~20 s (see time series and supershear reference in Figure S11 of Supporting Information S1). The average rupture speed for the Craig earthquake inferred by us agrees with the finite-fault modeling results (4–5 km/s) of Aderhold and





Figure 10. Rupture length scaling relations for short-period back-projection based estimates. (a) Normal, (b) thrust, and (c) strike-slip earthquakes. The black line shows the back-projection relationships (solid line) and the 95% of confidence interval predicted (dashed line). The squares, circles, and triangles present the data distribution for each faulting type. The gray circle shows the 2021 South of Sandwich Islands earthquake not included in the regression. Blue and red lines show the well-known scaling relationships of Blaser et al. (2010) and Wells and Coppersmith (1994) for comparison.

Abercrombie (2015). Previously, Yue et al. (2013) provided faster peak velocity estimates from Sg and Sn arrivals (>4.5 km/s) and finite-fault inversion (5.5–6 km/s), but as we do not consider time-variable propagation rates, these observations are not in conflict. The 2013 Balochistan event propagated at a speed of 3.3 km/s (Figure S29 in Supporting Information S1), representing 98% of the shear wave speed, that is, around the shear velocity but (very likely) faster than Rayleigh waves. Additionally, the 2018 North of Honduras (95%V_s; Figure S51 in Supporting Information S1) and 2020 Caribbean (104%V_s; Figure S59 in Supporting Information S1) earthquakes propagated at "likely" supershear (95–105%V_s; by assuming expected rupture velocity errors in the estimates). For the Caribbean earthquake, Tadapansawut et al. (2021) inferred supershear rupture fronts with peak velocities larger than 5 km/s from finite-fault modeling, that is, significantly faster than our average-based estimates.

For average velocity estimates based on manually and automatically determined rupture end times, please consult Tables S4–S5 in Supporting Information S1 and Data Sets S2–S3. Additionally, refer to Figure S80 in Supporting Information S1 for an illustrative example of the estimation of average velocities based on manually determined rupture end times. Overall, the average rupture speeds are approximately 70% and 65% of their respective maximum values. This is in relation to the manual and automatic rupture end times, respectively. The decrease in estimated rupture speed based on the averages could potentially lead to an underestimation or complicate the evaluation of supershear ruptures. This effect may be particularly pronounced when a rupture transitions from subshear to supershear, as observed in the 2013 Craig earthquake (shifting from 4.6 km/s or $137\%V_s$ to 2.9 km/s or $86\%V_s$; Table S4 in Supporting Information S1). Therefore, we preferred maximum velocity values over averages.

3.5.3. Rupture Length

The rupture length estimates from backprojections were used to derive magnitude-length scaling relations for each faulting slip type (normal, thrust, and strike-slip), based on the logarithm of rupture length and moment magnitude (Figure 10 and Table S6 in Supporting Information S1). Due to the composite nature of the earthquake and variability in moment estimates, the 2021 Sandwich Islands event was not included in the regression for thrust magnitude-length scaling relation. Thrust and normal earthquakes showed similar magnitude-length dependencies, while strike-slip earthquakes had a large data dispersion on average.

We additionally compared the newly short-period scaling relations to established scaling relations (e.g., Blaser et al., 2010; Wells & Coppersmith, 1994) derived from aftershocks, geodetic modeling, and empirical



relationships (Figure 10, Table S6 in Supporting Information S1). For normal faulting earthquakes (Figure 10a), our scaling relation is very similar to those derived from both Wells and Coppersmith (1994) and Blaser et al. (2010). For thrust earthquakes (Figure 10b), our scaling relation predicts shorter rupture lengths than Wells and Coppersmith (1994) but close to the predictions of Blaser et al. (2010), which is based on a much larger database. Lengths for strike-slip earthquakes presented a large dispersion, with the regression pointing to slower growth of rupture length with magnitude than either Blaser et al. (2010) or Wells and Coppersmith (1994) but still within the expected, albeit large, uncertainty range (Figure 10c). The first limitation here is the sparsity of strike-slip events with $M_W > \sim 8$. Second, several of the large strike-slip earthquakes were complex and included multiple conjugate faults, leading to large ambiguity in the interpretation of the rupture length as determined by our algorithm (as the cumulative fault length ruptured will be much larger than this estimate). This pattern is often found for intraplate earthquakes in the oceanic lithosphere, for example, the great 2012 Wharton Basin (e.g., Duputel et al., 2012; Hill et al., 2015; Meng, Ampuero, Stock, et al., 2012; Satriano et al., 2012) and 2018 Gulf of Alaska (e.g., Krabbenhoeft et al., 2018; Lay et al., 2018; Ruppert et al., 2018) earthquakes, where backprojections were broadly compatible with the more comprehensive assessments for those events, for example, aftershocks, seafloor topography and/or finite fault modeling.

3.5.4. Aspect Ratio

We remind that manually determined rupture end times were preferred over automatic values. Therefore, the following analysis benefits from excluding potential outliers after the rupture ends. We also considered the full back-projected rupture, that is, we have not removed potential outliers from the middle part of the rupture.

We explored the relationship between aspect ratios and source complexity. Simple elongated ruptures were characterized by small aspect ratios (aspect ratio close to 0: very low sparsity in the rupture emissions), for example, the 2017 Komandorsky Islands (ratio ~0.19; Figure S49 in Supporting Information S1), 2016 Wharton Basin (ratio ~0.25; Figure S40 in Supporting Information S1), and 2011 Kermadec Islands (ratio ~0.17; Figure S19 in Supporting Information S1) earthquakes (all of them bilateral ruptures). In contrast, complex events with no straightforward interpretation of "rupture length" tended to have large values (aspect ratio close to 1: high sparsity in the rupture emissions). For strike-slip earthquakes, aspect ratios close to 1 often indicate the activation of multiple faults. Prominent examples are the 2012 Wharton Basin (ratio ~0.81; Figure S21 in Supporting Information S1) and Gulf of Alaska (ratio ~0.70; Figure S52 in Supporting Information S1) earthquakes that involved rupture of conjugate fault systems (e.g., Krabbenhoeft et al., 2018; Satriano et al., 2012).

It should be noted that some strike-slip earthquakes exhibit fluctuations in high-frequency rupture emissions that are not reflected in their small aspect ratio estimates due to their total rupture length. For instance, the 2018 North of Honduras (M_W 7.5) strike-slip earthquake presented a low ratio (ratio ~0.29; Figure S51 in Supporting Information S1). This low ratio indicates an elongated rupture, as observed in the back-projection (Figure S51 in Supporting Information S1). However, it is also possible to observe NS fluctuations in the rupture emissions that are not reflected in the overall ratio due to the large EW extent of the rupture (rupture length: 143 km; Figure S51c in Supporting Information S1). Aspect ratios for sub-segments of the North of Honduras rupture would indicate higher values relative to the total evaluation of the earthquake extent, likely attributed to fluctuations in the high-frequency radiation, rather than having a physical meaning.

Supershear earthquakes were found to be associated with the most mature parts of the rupture zone (Perrin et al., 2016). Since fault length can be considered a proxy for the fault maturity (e.g., Manighetti et al., 2007; Perrin et al., 2016), back-projected speeds in the supershear range combined with linear faults (small aspect ratios < \sim 0.3), indicate a large degree in fault maturity, for example, Palu (ratio \sim 0.21; Figure S54 in Supporting Information S1) and Craig (ratio \sim 0.31; Figure S26 in Supporting Information S1) supershear earthquakes. Not all strike-slip earthquakes with a small aspect ratio were supershear. For example, the 2016 Wharton Basin is a sub-shear event with a small aspect ratio (ratio \sim 0.25; Figure S40 in Supporting Information S1). For thrust earthquakes, the aspect ratio also quantified complexity. However, the complexity usually does not arise from a rupture on different faults but from the distribution of high-frequency emissions on the megathrust, for example, the 2015 Illapel earthquake and its balanced rupture pattern (ratio \sim 0.84; Figure S39 in Supporting Information S1).

Regardless of the faulting slip type, a compact rupture can also result in aspect ratios close to 1 due to the greater importance of variance in the estimated locations of emission points for smaller ruptures, for example, the 2021 Loyalty Islands (ratio ~0.84; Figure S62 in Supporting Information S1) and 2014 Solomon Islands (ratio ~0.61;



Figure S34 in Supporting Information S1) earthquakes. Therefore, rupture complexity is linked to aspect ratios > ~ 0.5 , but such a large value does not necessarily imply complexity but can also indicate compactness, particularly for earthquakes with $M_W < \sim 8$.

4. Conclusions

We present a catalog of short-period rupture histories for 56 large earthquakes (0.5–2.0 Hz; $M_W \ge 7.5$; 01/2010–12/2022) based on a new implementation of the teleseismic back-projection method. The approach includes multiple arrays and combines P and pP waveforms for earthquakes deeper than 40 km, which also considers the radiation pattern, that is, in selecting only those arrays where the pP depth phase is expected to have a significant amplitude. Based on the backprojections, rupture length, directivity, speed, and aspect ratio were estimated algorithmically. The main findings are as follows:

- We found differences in rupture patterns between finite fault slip models and short-period emissions for subduction megathrust earthquakes. We confirmed the preference for short-period seismic energy to be emitted downdip of the main slip asperity (as previously reported in the literature), but additionally identified many examples of balanced rupture patterns around the main asperity, including updip short-period ruptures.
- Short-period ruptures for megathrust earthquakes, for example, balanced patterns around slip patches, might result from the high-stress gradient around asperities and seismogenic barriers (strong gradients in the slip are needed to generate high-frequency emissions).
- 3. Earthquake rupture speeds were consistently in the sub-Rayleigh regime for thrust and normal faulting earthquakes, with a median of 56% and 60% of the shear wave speed, respectively. Strike-slip earthquakes showed the greatest variability with a median of 78% and a range between 37% and 137% of the shear wave speed. The 2013 Craig ($137\%V_s$) and 2018 Palu ($125\%V_s$) events propagated in the unstable supershear range. The 2013 Balochistan ($98\%V_s$), 2018 North of Honduras ($95\%V_s$), and 2020 Caribbean ($104\%V_s$) earthquakes were "likely" supershear ($95-105\%V_s$; by assuming expected rupture velocity errors in the estimates) and in any case faster than Rayleigh waves.
- 4. We presented new scaling relations from short-period backprojections comparable to finite difference methods. Thrust and normal earthquakes showed a similar magnitude-length relationship compared to established "long-period" relations. Strike-slip events presented a large dispersion but still within the expected uncertainty of previously published scaling relations. Limitations for strike-slip earthquakes were the lack of events with a large moment magnitude ($M_W > \sim 8$) and the difficulty of interpreting rupture lengths for ruptures on complex conjugate fault systems.

Whereas overall, the method is suitable for near-real-time processing, the array selection and the definition of secondary search grids need to be done manually. Furthermore, the automatic determination of the end of the rupture is not always robust and needs to be confirmed manually. This paper has analyzed the fault rupture histories for 56 earthquakes, but we intend to continue to analyze future earthquakes with $M_W \ge 7.5$ and depths less than 200 km.

Data Availability Statement

The back-projection catalog is available in Supporting Information S1 as map views and in machine-readable format (Data Set S1), including the earthquake parameters derived from back-projection (Data Set S2) and parameter estimates based on automatic rupture end times (Data Set S3). The data files used in this paper are available at Vera et al. (2024): GFZ Data Services (https://doi.org/10.5880/GFZ.2.4.2024.001). Earthquake source parameters obtained from the CSN, GEOFON, ISC, SSN, USGS-NEIC catalogs, and Sippl et al. (2018), Lay et al. (2013), Nocquet et al. (2017), Bilek et al. (2011), Zhao et al. (2011), and Adhikari et al. (2015); see Tables S1–S3 in Supporting Information S1 for additional details. Fault geometry for the 2013 Balochistan (Figure S29 in Supporting Information S1) and 2016 Kaikoura (Figure S43 in Supporting Information S1) earthquakes extracted from Avouac et al. (2014) and Hamling et al. (2017). Major plate boundaries from Styron and Pagani (2020). Finite fault slip solutions for the 2015 Gorkha (Figure S37 in Supporting Information S1) and 2017 Chiapas (Figure S50 in Supporting Information S1) earthquakes from Yagi and Okuwaki (2015) and Okuwaki and Yagi (2017). Hi-net data was provided by the National Research Institute for Earth Science and Disaster Prevention (NIED). NIED/Hi-net data was obtained with the HinetPy Python package (Tian, 2020). The



European Integrated Data Archive and IRIS Data Management Centre were used to access waveforms. We backprojected data from the AR, AU, BA, C, DK, EP, ER, GG, HW, IM, LB, MU, MY, NM, PB, PL, SC, SI, X4, ZA and ZP FDSN networks, and those administrated by Frank Vernon (1982), California Institute of Technology and United States Geological Survey Pasadena (1926), IRIS Transportable Array (2003), Albuquerque Seismological Laboratory (ASL)/USGS (1990), Mike Brudzinski (2007), Penn State University (2004), A. N. Douglas Wiens (2007b), Jay Pulliam (2008), University of Utah (1962), Northern California Earthquake Data Center (2014), Arizona Geological Survey (2009), Albuquerque Seismological Laboratory (ASL)/USGS (1980, 2003), UC San Diego (2013), University of Washington (1963), Vernon and BenZion (2010), Steve Gao (2012), Utrecht University (UU Netherlands) (1983), Hansen (2012), Natural Resources Canada (NRCAN Canada) (1975), Lamont Doherty Earth Observatory (LDEO), Columbia University (1970), Australian National University (ANU, Australia) (2011), Universidad De Chile (2013), P. B. Douglas Wiens (2014), University of Puerto Rico (1986), GFZ German Research Centre For Geosciences and Institut Des Sciences De L'Univers-Centre National De La Recherche CNRS-INSU (2006), Frank Vernon (2014), Ghosh (2014), Alaska Earthquake Center (1987), NOAA National Oceanic and Atmospheric Administration (USA) (1967), Alaska Volcano Observatory/USGS (1988), Conder (2013), Maureen Long (2013), Nettles (2014), Albuquerque Seismological Laboratory (ASL)/USGS (1993), Ohio Geological Survey (1999), Jason Hebert (2016), GEOFON Data Centre (1993), Federal Institute for Geosciences and Natural Resources (1976), INGV Seismological Data Centre (1997), Leipzig University (2001), Institute of Geophysics (1973), RESIF (1995), Jena (2009), National Institute For Earth Physics (NIEP Romania) (1994), OGS (Istituto Nazionale Di Oceanografia E Di Geofisica Sperimentale) (2016), ZAMG-Zentralanstalt Für Meterologie Und Geodynamik (1987), Department of Earth and Environmental Sciences, Geophysical Observatory, University of Munchen (2001), Derek Schutt (2015), Roger Hansen (2005), Russo (2011), Institut De Physique Du Globe De Paris (IPGP) and Ecole Et Observatoire Des Sciences De La Terre De Strasbourg (EOST) (1982), Klemperer (2011), W. S. Lee (2013), Liang et al. (2016), Institute of Geophysics China Earthquake Administration (IGPCEA) (2007), University of Genoa (1967), MedNet Project Partner Institutions (1988), Lara Wagner (2010), Albuquerque Seismological Laboratory (ASL)/ USGS (1994), Oklahoma Geological Survey (1978), Gary Pavlis (2011), Wagner (2012), Charles Langston (2011), William Menke (2012), S. V. D. Lee et al. (2011), University of Oregon (1990), Albuquerque Seismological Laboratory (ASL)/USGS (1992), Simon Klemperer (2010), Kövesligethy Radó Seismological Observatory (Geodetic and Geophysical Institute, Research Centre For Astronomy and Earth Sciences, Hungarian Academy of Sciences (MTA CSFK GGI KRSZO)) (1992), Geoffrey Abers (2010), Leroy et al. (2009), Jim Gaherty (2013), Derek Keir (2014), Barruol et al. (2017), Institute Earth Sciences "Jaume Almera" CSIC (ICTJA Spain) (2007), Helffrich and Fonseca (2011), Bascou and Barruol (2015), Albuquerque Seismological Laboratory (ASL)/USGS (1988), A. N. Douglas Wiens (2007a), Kandilli Observatory and Earthquake Research Institute, Boğaziçi University (1971), OGS (Istituto Nazionale Di Oceanografia E Di Geofisica Sperimentale) and University of Trieste (2002), Michael Wysession (2011), Susan Beck (2010), Bureau of Economic Geology (2016), Bruce Beaudoin (2011), and Qian and Mechie (2012).

Acknowledgments

This research was funded by the National Agency for Research and Development (ANID). Scholarship Program: Doctorado Becas Chile (2017-72180166). Open Access funding enabled and organized by Projekt DEAL.

References

Aderhold, K., & Abercrombie, R. (2015). Seismic rupture on an oceanic–continental plate boundary: Strike-slip earthquakes along the Queen Charlotte–Fairweather Fault. *Bulletin of the Seismological Society of America*, *105*(2B), 1129–1142. https://doi.org/10.1785/0120140227

Adhikari, L., Gautam, U., Koirala, B., Bhattarai, M., Kandel, T., Gupta, R., et al. (2015). The aftershock sequence of the 2015 April 25 Gorkha– Nepal earthquake. *Geophysical Supplements to the Monthly Notices of the Royal Astronomical Society*, 203(3), 2119–2124. https://doi.org/10. 1093/gji/ggv412

Agurto-Detzel, H., Font, Y., Charvis, P., Régnier, M., Rietbrock, A., Ambrois, D., et al. (2019). Ridge subduction and afterslip control aftershock distribution of the 2016 Mw 7.8 Ecuador earthquake. *Earth and Planetary Science Letters*, 520, 63–76. https://doi.org/10.1016/j.epsl.2019. 05.029

Aki, K., & Richards, P. G. (2002). Quantitative seismology.

Alaska Earthquake Center, U. O. A. F. (1987). Alaska Regional Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/AK

Alaska Volcano Observatory/USGS. (1988). Alaska Volcano Observatory [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/AV

Albuquerque Seismological Laboratory (ASL)/USGS. (1980). US Geological Survey Networks [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/GS

Albuquerque Seismological Laboratory (ASL)/USGS. (1988). Global Seismograph Network (GSN - IRIS/USGS) [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/IU

Albuquerque Seismological Laboratory (ASL)/USGS. (1990). United States National Seismic Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/US



- Albuquerque Seismological Laboratory (ASL)/USGS. (1992). New China Digital Seismograph Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/IC
- Albuquerque Seismological Laboratory (ASL)/USGS. (1993). Global Telemetered Seismograph Network (USAF/USGS) [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/GT
- Albuquerque Seismological Laboratory (ASL)/USGS. (1994). New England Seismic Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/NE
- Albuquerque Seismological Laboratory (ASL)/USGS. (2003). Intermountain West Seismic Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/IW
- An, C., & Meng, L. (2016). Application of array backprojection to tsunami prediction and early warning. *Geophysical Research Letters*, 43(8), 3677–3685. https://doi.org/10.1002/2016gl068786
- Andrews, D. (1976). Rupture velocity of plane strain shear cracks. Journal of Geophysical Research, 81(32), 5679–5687. https://doi.org/10.1029/jb081i032p05679
- Archuleta, R. J. (1984). A faulting model for the 1979 Imperial Valley earthquake. Journal of Geophysical Research, 89(B6), 4559–4585. https:// doi.org/10.1029/jb089ib06p04559
- Arizona Geological Survey. (2009). Arizona Broadband Seismic Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/AE
- Australian National University (ANU, Australia). (2011). Australian Seismometers in Schools [Dataset]. Australian Passive Seismic Server Australian National University. https://doi.org/10.7914/SN/S1
- Avouac, J.-P., Ayoub, F., Wei, S., Ampuero, J.-P., Meng, L., Leprince, S., et al. (2014). The 2013, Mw 7.7 Balochistan earthquake, energetic strike-slip reactivation of a thrust fault. *Earth and Planetary Science Letters*, 391, 128–134. https://doi.org/10.1016/j.epsl.2014.01.036
- Bao, H., Ampuero, J.-P., Meng, L., Fielding, E. J., Liang, C., Milliner, C. W., et al. (2019). Early and persistent supershear rupture of the 2018 magnitude 7.5 Palu earthquake. *Nature Geoscience*, 12(3), 200–205. https://doi.org/10.1038/s41561-018-0297-z
- Barruol, G., Sigloch, K., RHUM-RUM Group, & RESIF. (2017). RHUM-RUM experiment, 2011-2015, code YV (Réunion Hotspot and Upper Mantle – Réunion's Unterer Mantel) funded by ANR, DFG, CNRS-INSU, IPEV, TAAF, instrumented by DEPAS, INSU-OBS, AWI and the Universities of Muenster, Bonn, La Réunion [Dataset]. RESIF - Réseau Sismologique et Géodésique Français. https://doi.org/10.15778/RESIF. YV2011
- Bascou, J., & Barruol, G. (2015). ArLiTA: Architecture de la Lithosphère de Terre Adélie, Antarctique, code 1A, funded and instrumented by IPEV, CNRS-INSU, RESIF-SISMOB, Laboratoire Magmas et Volcans (UMR 6524), Université de Saint Etienne, Université de La Réunion, IPGP [Dataset]. RESIF - Réseau Sismologique et Géodésique Français. https://doi.org/10.15778/RESIF.1A2009
- Bilek, S. L., Engdahl, E. R., DeShon, H. R., & El Hariri, M. (2011). The 25 October 2010 Sumatra tsunami earthquake: Slip in a slow patch. Geophysical Research Letters, 38(14), L14306. https://doi.org/10.1029/2011gl047864
- Bird, P. (2003). An updated digital model of plate boundaries. *Geochemistry, Geophysics, Geosystems*, 4(3), 1027. https://doi.org/10.1029/2001gc000252
- Blaser, L., Krüger, F., Ohrnberger, M., & Scherbaum, F. (2010). Scaling relations of earthquake source parameter estimates with special focus on subduction environment. *Bulletin of the Seismological Society of America*, 100(6), 2914–2926. https://doi.org/10.1785/0120100111
- Bouchon, M., Bouin, M.-P., Karabulut, H., Toksöz, M. N., Dietrich, M., & Rosakis, A. J. (2001). How fast is rupture during an earthquake? New insights from the 1999 Turkey earthquakes. *Geophysical Research Letters*, 28(14), 2723–2726. https://doi.org/10.1029/2001gl013112
- Bouchon, M., Karabulut, H., Bouin, M.-P., Schmittbuhl, J., Vallée, M., Archuleta, R., et al. (2010). Faulting characteristics of supershear earthquakes. *Tectonophysics*, 493(3–4), 244–253. https://doi.org/10.1016/j.tecto.2010.06.011
- Bouchon, M., & Vallée, M. (2003). Observation of long supershear rupture during the magnitude 8.1 Kunlunshan earthquake. *Science*, 301(5634), 824–826. https://doi.org/10.1126/science.1086832
- Bruce Beaudoin, T. P. (2011). Testing TA & FA vaults and directly buried sensor (3T) [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/YE_2011
- Bureau of Economic Geology, T. U. O. T. A. A. (2016). Texas Seismological Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/TX
- Burridge, R. (1973). Admissible speeds for plane-strain self-similar shear cracks with friction but lacking cohesion. Geophysical Journal International, 35(4), 439–455. https://doi.org/10.1111/j.1365-246x.1973.tb00608.x
- Burridge, R., Conn, G., & Freund, L. (1979). The stability of a rapid mode II shear crack with finite cohesive traction. Journal of Geophysical Research, 84(B5), 2210–2222. https://doi.org/10.1029/jb084ib05p02210
- California Institute of Technology and United States Geological Survey Pasadena. (1926). Southern California Seismic Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/CI
- Charles Langston, H. D. (2011). Northern Embayment Lithospheric Experiment [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/ZL_2011
- Conder, J. (2013). Wabash Valley Seismic Zone [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/ SN/6E_2013
- Crotwell, H. P., Owens, T. J., & Ritsema, J. (1999). The TauP Toolkit: Flexible seismic travel-time and ray-path utilities. Seismological Research Letters, 70(2), 154–160. https://doi.org/10.1785/gssrl.70.2.154
- Crowell, B. W., & Melgar, D. (2020). Slipping the Shumagin Gap: A kinematic coseismic and early afterslip model of the Mw 7.8 Simeonof Island, Alaska, earthquake. *Geophysical Research Letters*, 47(19), e2020GL090308. https://doi.org/10.1029/2020gl090308
- D'Amico, S., Koper, K. D., Herrmann, R. B., Akinci, A., & Malagnini, L. (2010). Imaging the rupture of the Mw 6.3 April 6, 2009 L'Aquila, Italy earthquake using back-projection of teleseismic P-waves. *Geophysical Research Letters*, 37(3), L03301. https://doi.org/10.1029/2009gl042156
- Das, S., & Aki, K. (1977). A numerical study of two-dimensional spontaneous rupture propagation. *Geophysical Journal International*, 50(3), 643–668. https://doi.org/10.1111/j.1365-246x.1977.tb01339.x
- Department of Earth and Environmental Sciences, Geophysical Observatory, University of Munchen. (2001). BayernNetz [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/BW
- Derek Keir, C. E. (2014). Southern Lake Tanganyika experiment [Dataset]. International Federation of Digital Seismograph Networks. https://doi. org/10.7914/SN/ZV_2014
- Derek Schutt, R. A. (2015). The Mackenzie Mountains Transect: Active Deformation from Margin to Craton [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/7C_2015
- Douglas Wiens, A. N. (2007a). A Broadband Seismic Experiment to Image the Lithosphere beneath the Gamburtsev Mountains, East Antarctica [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/ZM_2007



- Douglas Wiens, A. N. (2007b). IPY POLENET-Antarctica: Investigating links between geodynamics and ice sheets [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/YT_2007
- Douglas Wiens, P. B. (2014). Collaborative Research: Dynamic Response of the Ross Ice Shelf to Wave-Induced Vibrations and Collaborative Research: Mantle Structure and Dynamics of the Ross Sea from a Passive Seismic Deployment on the Ross Ice Shelf [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/XH_2014
- Dunham, E. M., & Archuleta, R. J. (2004). Evidence for a supershear transient during the 2002 Denali fault earthquake. Bulletin of the Seismological Society of America, 94(6B), S256–S268. https://doi.org/10.1785/0120040616
- Duputel, Z., Kanamori, H., Tsai, V. C., Rivera, L., Meng, L., Ampuero, J.-P., & Stock, J. M. (2012). The 2012 Sumatra great earthquake sequence. Earth and Planetary Science Letters, 351, 247–257. https://doi.org/10.1016/j.epsl.2012.07.017
- Federal Institute for Geosciences and Natural Resources. (1976). German Regional Seismic Network (GRSN) [Dataset]. Bundesanstalt für Geowissenschaften und Rohstoffe. https://doi.org/10.25928/MBX6-HR74
- Feng, T., Meng, L., & Huang, H. (2020). Detecting Offshore Seismicity: Combining Backprojection Imaging and Matched-Filter Detection. Journal of Geophysical Research: Solid Earth, 125(8), e2020JB019599. https://doi.org/10.1029/2020jb019599
- Frank Vernon, U. S. D. (1982). ANZA Regional Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/AZ
- Frank Vernon, U. S. D. (2014). Piñon Flats Observatory (PFO) Array [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/PY
- Gary Pavlis, H. G. (2011). Ozark Illinois Indiana Kentucky (OIINK) Flexible Array Experiment [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/XO_2011
- Geoffrey Abers, J. G. (2010). How is Rifting Exhuming the Youngest HP/UHP Rocks on Earth? [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/HU
- GEOFON Data Centre. (1993). GEOFON Seismic Network [Dataset]. Deutsches GeoForschungsZentrum GFZ. https://doi.org/10.14470/ TR560404
- GFZ German Research Centre For Geosciences, & Institut Des Sciences De L'Univers-Centre National De La Recherche CNRS-INSU. (2006). IPOC Seismic Network [Dataset]. Integrated Plate boundary Observatory Chile - IPOC. https://doi.org/10.14470/PK615318
- Ghosh, A. (2014). Aleutian Array of Arrays [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/9C_2014
- Hamling, I. J., Hreinsdóttir, S., Clark, K., Elliott, J., Liang, C., Fielding, E., et al. (2017). Complex multifault rupture during the 2016 Mw 7.8 Kaikõura earthquake, New Zealand. Science, 356(6334), eaam7194. https://doi.org/10.1126/science.aam7194
- Hansen, S. (2012). Transantarctic Mountains Northern Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi. org/10.7914/SN/ZJ_2012
- Hayes, G. P., Wald, D. J., & Johnson, R. L. (2012). Slab1.0: A three-dimensional model of global subduction zone geometries. *Journal of Geophysical Research*, 117(B1), B01302. https://doi.org/10.1029/2011jb008524
- Heidarzadeh, M., Murotani, S., Satake, K., Takagawa, T., & Saito, T. (2017). Fault size and depth extent of the Ecuador earthquake (Mw 7.8) of 16 April 2016 from teleseismic and tsunami data. *Geophysical Research Letters*, 44(5), 2211–2219. https://doi.org/10.1002/2017gl072545
- Helffrich, G., & Fonseca, J. F. B. D. (2011). Mozambique Rift Tomography [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/6H_2011
- Hill, E. M., Borrero, J. C., Huang, Z., Qiu, Q., Banerjee, P., Natawidjaja, D. H., et al. (2012). The 2010 Mw 7.8 Mentawai earthquake: Very shallow source of a rare tsunami earthquake determined from tsunami field survey and near-field GPS data. *Journal of Geophysical Research*, 117(B6), B06402. https://doi.org/10.1029/2012jb009159
- Hill, E. M., Yue, H., Barbot, S., Lay, T., Tapponnier, P., Hermawan, I., et al. (2015). The 2012 Mw 8.6 Wharton Basin sequence: A cascade of great earthquakes generated by near-orthogonal, young, oceanic mantle faults. *Journal of Geophysical Research: Solid Earth*, 120(5), 3723–3747. https://doi.org/10.1002/2014jb011703
- Iinuma, T., Hino, R., Kido, M., Inazu, D., Osada, Y., Ito, Y., et al. (2012). Coseismic slip distribution of the 2011 off the Pacific coast of Tohoku Earthquake (M9.0) refined by means of seafloor geodetic data. *Journal of Geophysical Research*, 117(B7), B07409. https://doi.org/10.1029/ 2012jb009186
- INGV Seismological Data Centre. (1997). Rete Sismica Nazionale (RSN) [Dataset]. Istituto Nazionale di Geofisica e Vulcanologia (INGV), Italy. https://doi.org/10.13127/SD/X0FXNH7QFY
- Institut De Physique Du Globe De Paris (IPGP), & Ecole Et Observatoire Des Sciences De La Terre De Strasbourg (EOST). (1982). GEOSCOPE, French Global Network of broad band seismic stations [Dataset]. Institut de physique du globe de Paris (IPGP), Université de Paris. https://doi. org/10.18715/GEOSCOPE.G
- Institute Earth Sciences "Jaume Almera" CSIC (ICTJA Spain). (2007). IberArray [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/IB
- Institute of Geophysics, A. O. S. O. T. C. R. (1973). Czech Regional Seismic Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/CZ
- Institute of Geophysics China Earthquake Administration (IGPCEA). (2007). Data Management Centre of China National Seismic Network at Institute of Geophysics, CEA [Dataset]. Data Management Centre of China National Seismic Network at Institute of Geophysics, CEA. https:// doi.org/10.7914/SN/CB
- IRIS Transportable Array. (2003). USArray Transportable Array [Dataset]. International Federation of Digital Seismograph Networks. https://doi. org/10.7914/SN/TA
- Ishii, M., Shearer, P. M., Houston, H., & Vidale, J. E. (2005). Extent, duration and speed of the 2004 Sumatra–Andaman earthquake imaged by the Hi-Net array. *Nature*, 435(7044), 933–936. https://doi.org/10.1038/nature03675
- Ishii, M., Shearer, P. M., Houston, H., & Vidale, J. E. (2007). Teleseismic P wave imaging of the 26 December 2004 Sumatra-Andaman and 28 March 2005 Sumatra earthquake ruptures using the Hi-net array. *Journal of Geophysical Research*, 112(B11), B11307. https://doi.org/10.1029/ 2006jb004700
- Jason Hebert, D. G. (2016). Interim Broadband Monitoring of the Mount Erebus Volcano [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/2H_2016
- Jay Pulliam, S. G. (2008). Seismic Investigation of Edge Driven Convection Associated with the Rio Grande Rift [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/XR_2008
- Jena, F. S. U. (2009). Thüringer Seismologisches Netz (TSN) [Dataset]. International Federation of Digital Seismograph Networks. https://doi. org/10.7914/SN/TH



Jim Gaherty, C. E. (2013). Study of Extension and maGmatism in Malawi aNd Tanzania [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/YQ_2013

Kanamori, H. (1981). The nature of seismicity patterns before large earthquakes. Earthquake Prediction.

- Kandilli Observatory and Earthquake Research Institute, Boğaziçi University. (1971). Bogazici University Kandilli Observatory and Earthquake Research Institute [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/KO
- Kelleher, J., & Savino, J. (1975). Distribution of seismicity before large strike slip and thrust-type earthquakes. *Journal of Geophysical Research*, 80(2), 260–271. https://doi.org/10.1029/jb080i002p00260
- Kennett, B., & Engdahl, E. (1991). Traveltimes for global earthquake location and phase identification. Geophysical Journal International, 105(2), 429–465. https://doi.org/10.1111/j.1365-246x.1991.tb06724.x
- Kiser, E., & Ishii, M. (2013). Hidden aftershocks of the 2011 Mw 9.0 Tohoku, Japan earthquake imaged with the backprojection method. Journal of Geophysical Research: Solid Earth, 118(10), 5564–5576. https://doi.org/10.1002/2013jb010158
- Kiser, E., & Ishii, M. (2017). Back-projection imaging of earthquakes. Annual Review of Earth and Planetary Sciences, 45(1), 271–299. https:// doi.org/10.1146/annurev-earth-063016-015801
- Kiser, E., Ishii, M., Langmuir, C. H., Shearer, P., & Hirose, H. (2011). Insights into the mechanism of intermediate-depth earthquakes from source properties as imaged by back projection of multiple seismic phases. *Journal of Geophysical Research*, 116(B6), B06310. https://doi.org/10. 1029/2010jb007831
- Klemperer, S. (2011). Passive seismic study of a magma-dominated rift: The Salton Trough [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/XD_2011
- Koper, K. D., Hutko, A., & Lay, T. (2011). Along-dip variation of teleseismic short-period radiation from the 11 March 2011 Tohoku earthquake (Mw 9.0). Geophysical Research Letters, 38(21), L21309. https://doi.org/10.1029/2011g1049689
- Koper, K. D., Hutko, A. R., Lay, T., & Sufri, O. (2012). Imaging short-period seismic radiation from the 27 February 2010 Chile (Mw 8.8) earthquake by back-projection of P, PP, and PKIKP waves. *Journal of Geophysical Research*, 117(B2), B02308. https://doi.org/10.1029/ 2011jb008576
- Kövesligethy Radó Seismological Observatory (Geodetic and Geophysical Institute, Research Centre For Astronomy and Earth Sciences, Hungarian Academy of Sciences (MTA CSFK GGI KRSZO)). (1992). Hungarian National Seismological Network [Dataset]. Deutsches GeoForschungsZentrum GFZ. https://doi.org/10.14470/UH028726
- Krabbenhoeft, A., von Huene, R., Miller, J. J., Lange, D., & Vera, F. (2018). Strike-slip 23 January 2018 Mw 7.9 Gulf of Alaska rare intraplate earthquake: Complex rupture of a fracture zone system. *Scientific Reports*, 8(1), 1–9.
- Krüger, F., & Ohrnberger, M. (2005). Tracking the rupture of the Mw= 9.3 Sumatra earthquake over 1,150 km at teleseismic distance. Nature, 435(7044), 937–939. https://doi.org/10.1038/nature03696
- Lamont Doherty Earth Observatory (LDEO), Columbia University. (1970). Lamont-Doherty Cooperative Seismographic Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/LD
- Lara Wagner, S. B. (2010). PerU Lithosphere and Slab Experiment [Dataset]. International Federation of Digital Seismograph Networks. https:// doi.org/10.7914/SN/ZD_2010
- Laske, G., Masters, G., Ma, Z., & Pasyanos, M. (2013). Update on CRUST1. 0—A 1-degree global model of Earth's crust. In *Geophysical Research Abstracts* (Vol. 15, p. 2658).
- Lay, T., Ammon, C., Kanamori, H., Koper, K., Sufri, O., & Hutko, A. (2010). Teleseismic inversion for rupture process of the 27 February 2010 Chile (Mw 8.8) earthquake. *Geophysical Research Letters*, 37(13), L13301. https://doi.org/10.1029/2010gl043379
- Lay, T., Ammon, C. J., Kanamori, H., Yamazaki, Y., Cheung, K. F., & Hutko, A. R. (2011). The 25 October 2010 Mentawai tsunami earthquake (Mw 7.8) and the tsunami hazard presented by shallow megathrust ruptures. *Geophysical Research Letters*, 38(6), L06302. https://doi.org/10. 1029/2010gl046552
- Lay, T., Kanamori, H., Ammon, C. J., Koper, K. D., Hutko, A. R., Ye, L., et al. (2012). Depth-varying rupture properties of subduction zone megathrust faults. *Journal of Geophysical Research*, 117(B4), B04311. https://doi.org/10.1029/2011jb009133
- Lay, T., Ye, L., Ammon, C. J., & Kanamori, H. (2017). Intraslab rupture triggering megathrust rupture coseismically in the 17 December 2016 Solomon Islands Mw 7.9 earthquake. *Geophysical Research Letters*, 44(3), 1286–1292. https://doi.org/10.1002/2017g1072539
- Lay, T., Ye, L., Bai, Y., Cheung, K. F., & Kanamori, H. (2018). The 2018 Mw 7.9 Gulf of Alaska earthquake: Multiple fault rupture in the Pacific plate. *Geophysical Research Letters*, 45(18), 9542–9551. https://doi.org/10.1029/2018gl079813
- Lay, T., Ye, L., Kanamori, H., Yamazaki, Y., Cheung, K. F., Kwong, K., & Koper, K. D. (2013). The October 28, 2012 Mw 7.8 Haida Gwaii underthrusting earthquake and tsunami: Slip partitioning along the Queen Charlotte Fault transpressional plate boundary. *Earth and Planetary Science Letters*, 375, 57–70. https://doi.org/10.1016/j.epsl.2013.05.005
- Lee, S.-J., Lin, T.-C., Feng, K.-F., & Liu, T.-Y. (2018). Composite megathrust rupture from deep interplate to trench of the 2016 Solomon Islands earthquake. *Geophysical Research Letters*, 45(2), 674–681. https://doi.org/10.1002/2017gl076347
- Lee, S. V. D., Wiens, D., Revenaugh, J., Frederiksen, A., & Darbyshire, F. (2011). Superior Province Rifting Earthscope Experiment [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/XI_2011
- Lee, W. S. (2013). Korea Polar Observation Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10. 7914/SN/KP
- Leipzig University. (2001). SXNET Saxon Seismic Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/ 10.7914/SN/SX
- Leroy, S., Keir, D., & Stuart, G. (2009). Young Conjugate Margins Lab in the Gulf of Aden [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/XW_2009
- Liang, C., Wu, J., & Wang, L. (2016). Broadband Array in Longmenshan Gap [Dataset]. Chengdu University of Technology. https://doi.org/10. 7914/SN/3J_2016
- Liu, Z., Song, C., Meng, L., Ge, Z., Huang, Q., & Wu, Q. (2017). Utilizing a 3D global P-wave tomography model to improve backprojection imaging: A case study of the 2015 Nepal earthquake. *Bulletin of the Seismological Society of America*, 107(5), 2459–2466. https://doi.org/10. 1785/0120170091
- Madariaga, R. (1977). High-frequency radiation from crack (stress drop) models of earthquake faulting. *Geophysical Journal International*, 51(3), 625–651. https://doi.org/10.1111/j.1365-246x.1977.tb04211.x
- Madariaga, R. (1983). High frequency radiation from dynamic earthquake. Annales Geophysicae, 1, 17.
- Manighetti, I., Campillo, M., Bouley, S., & Cotton, F. (2007). Earthquake scaling, fault segmentation, and structural maturity. *Earth and Planetary Science Letters*, 253(3–4), 429–438. https://doi.org/10.1016/j.epsl.2006.11.004
- Marty, S., Passelègue, F., Aubry, J., Bhat, H., Schubnel, A., & Madariaga, R. (2019). Origin of high-frequency radiation during laboratory earthquakes. *Geophysical Research Letters*, 46(7), 3755–3763. https://doi.org/10.1029/2018gl080519



- Maureen Long, P. W. (2013). Mid-Atlantic Geophysical Integrative Collaboration [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/7A_2013
- MedNet Project Partner Institutions. (1988). Mediterranean Very Broadband Seismographic Network (MedNet) [Dataset]. Istituto Nazionale di Geofisica e Vulcanologia (INGV). https://doi.org/10.13127/SD/FBBBTDTD6Q
- Melgar, D., Fan, W., Riquelme, S., Geng, J., Liang, C., Fuentes, M., et al. (2016). Slip segmentation and slow rupture to the trench during the 2015, Mw8.3 Illapel, Chile earthquake. *Geophysical Research Letters*, 43(3), 961–966. https://doi.org/10.1002/2015gl067369
- Meng, L., Ampuero, J.-P., Luo, Y., Wu, W., & Ni, S. (2012). Mitigating artifacts in back-projection source imaging with implications for frequency-dependent properties of the Tohoku-Oki earthquake. *Earth Planets and Space*, 64(12), 5–1109. https://doi.org/10.5047/eps.2012. 05.010
- Meng, L., Ampuero, J.-P., Stock, J., Duputel, Z., Luo, Y., & Tsai, V. (2012). Earthquake in a maze: Compressional rupture branching during the 2012 Mw 8.6 Sumatra earthquake. *Science*, 337(6095), 724–726. https://doi.org/10.1126/science.1224030
- Meng, L., Bao, H., Huang, H., Zhang, A., Bloore, A., & Liu, Z. (2018). Double pincer movement: Encircling rupture splitting during the 2015 Mw 8.3 Illapel earthquake. *Earth and Planetary Science Letters*, 495, 164–173. https://doi.org/10.1016/j.epsl.2018.04.057
- Meng, L., Huang, H., Bürgmann, R., Ampuero, J. P., & Strader, A. (2015). Dual megathrust slip behaviors of the 2014 Iquique earthquake sequence. Earth and Planetary Science Letters, 411, 177–187. https://doi.org/10.1016/j.epsl.2014.11.041
- Meng, L., Inbal, A., & Ampuero, J.-P. (2011). A window into the complexity of the dynamic rupture of the 2011 Mw 9 Tohoku-Oki earthquake. Geophysical Research Letters, 38(7), L00G07. https://doi.org/10.1029/2011gl048118
- Meng, L., Zhang, A., & Yagi, Y. (2016). Improving back projection imaging with a novel physics-based aftershock calibration approach: A case study of the 2015 Gorkha earthquake. *Geophysical Research Letters*, 43(2), 628–636. https://doi.org/10.1002/2015gl067034
- Metz, M., Vera, F., Ponce, A. C., Cesca, S., Babeyko, A., Dahm, T., et al. (2022). Seismic and tsunamigenic characteristics of a multimodal rupture of rapid and slow stages. *Journal of Geophysical Research: Solid Earth*, 127(11), e2022JB024646. https://doi.org/10.1029/ 2022jb024646
- Michael Wysession, D. W. (2011). Investigation of sources of intraplate volcanism using PASSCAL broadband instruments in Madagascar, the Comores, and Mozambique [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/XV_2011
- Mike Brudzinski, R. A. (2007). Resolving structural control of episodic tremor and slip along the length of Cascadia [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/YW_2007
- Mogi, K. (1969). Some feature of recent seismic activity in and near Japan (2): Activity before and after great earthquakes. *Bulletin of the Earthquake Research Institute, University of Tokyo*, 47, 395–417.
- Moreno, M., Li, S., Melnick, D., Bedford, J., Baez, J., Motagh, M., et al. (2018). Chilean megathrust earthquake recurrence linked to frictional contrast at depth. *Nature Geoscience*, 11(4), 285–290. https://doi.org/10.1038/s41561-018-0089-5
- Moreno, M., Melnick, D., Rosenau, M., Baez, J., Klotz, J., Oncken, O., et al. (2012). Toward understanding tectonic control on the Mw 8.8 2010 Maule Chile earthquake. *Earth and Planetary Science Letters*, 321, 152–165. https://doi.org/10.1016/j.epsl.2012.01.006
- National Institute For Earth Physics (NIEP Romania). (1994). Romanian Seismic Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/RO
- Natural Resources Canada (NRCAN Canada). (1975). Canadian National Seismograph Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/CN
- Neidell, N. S., & Taner, M. T. (1971). Semblance and other coherency measures for multichannel data. *Geophysics*, 36(3), 482–497. https://doi. org/10.1190/1.1440186
- Nettles, M. (2014). Understanding Precambrian to Present Assembly of Greenland [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/XF_2014
- NOAA National Oceanic and Atmospheric Administration (USA). (1967). National Tsunami Warning Center Alaska Seismic Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/AT
- Nocquet, J.-M., Jarrin, P., Vallée, M., Mothes, P., Grandin, R., Rolandone, F., et al. (2017). Supercycle at the Ecuadorian subduction zone revealed after the 2016 Pedernales earthquake. *Nature Geoscience*, 10(2), 145–149. https://doi.org/10.1038/ngeo2864
- Northern California Earthquake Data Center. (2014). Berkeley Digital Seismic Network (BDSN) [Dataset]. Northern California Earthquake Data Center. https://doi.org/10.7932/BDSN
- OGS (Istituto Nazionale Di Oceanografia E Di Geofisica Sperimentale). (2016). North-East Italy Seismic Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/OX
- OGS (Istituto Nazionale Di Oceanografia E Di Geofisica Sperimentale) and University of Trieste. (2002). North-East Italy Broadband Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/NI
- Ohio Geological Survey. (1999). Ohio Seismic Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/OH
- Oklahoma Geological Survey. (1978). Oklahoma Seismic Network [Dataset]. International Federation of Digital Seismograph Networks. https:// doi.org/10.7914/SN/OK
- Okuwaki, R., & Yagi, Y. (2017). Rupture process during the Mw 8.1 2017 Chiapas Mexico earthquake: Shallow intraplate normal faulting by slab bending. *Geophysical Research Letters*, 44(23), 11–816. https://doi.org/10.1002/2017gl075956
- Palo, M., Tilmann, F., Krueger, F., Ehlert, L., & Lange, D. (2014). High-frequency seismic radiation from Maule earthquake (Mw 8.8, 2010 February 27) inferred from high-resolution backprojection analysis. *Geophysical Journal International*, 199(2), 1058–1077. https://doi.org/10. 1093/gji/ggu311

Penn State University. (2004). AfricaArray [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/AF Perrin, C., Manighetti, I., Ampuero, J.-P., Cappa, F., & Gaudemer, Y. (2016). Location of largest earthquake slip and fast rupture controlled by

- along-strike change in fault structural maturity due to fault growth. Journal of Geophysical Research: Solid Earth, 121(5), 3666–3685. https:// doi.org/10.1002/2015jb012671
- Qian, H., & Mechie, J. (2012). Temporary seismological network in Longmenshan, LONGMEN (2012/2013) [Dataset]. GFZ Data Services. https://doi.org/10.14470/7S7567431325
- Rawlinson, N., & Kennett, B. L. (2004). Rapid estimation of relative and absolute delay times across a network by adaptive stacking. *Geophysical Journal International*, 157(1), 332–340. https://doi.org/10.1111/j.1365-246x.2004.02188.x
- RESIF. (1995). RESIF-RLBP French Broad-band network, RESIF-RAP strong motion network and other seismic stations in metropolitan France [Dataset]. RESIF Réseau Sismologique et géodésique Français. https://doi.org/10.15778/RESIF.FR
- Ricker, N. (1953). The form and laws of propagation of seismic wavelets. Geophysics, 18(1), 10-40. https://doi.org/10.1190/1.1437843
- Robinson, D., Das, S., & Searle, M. (2010). Earthquake fault superhighways. *Tectonophysics*, 493(3–4), 236–243. https://doi.org/10.1016/j.tecto. 2010.01.010



- Roger Hansen, G. P. (2005). Collaborative Research: St. Elias Erosion/Tectonics Project [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/XZ_2005
- Rössler, D., Krueger, F., Ohrnberger, M., & Ehlert, L. (2010). Rapid characterisation of large earthquakes by multiple seismic broadband arrays. *Natural Hazards and Earth System Sciences*, 10(4), 923–932. https://doi.org/10.5194/nhess-10-923-2010
- Ruppert, N., Rollins, C., Zhang, A., Meng, L., Holtkamp, S., West, M., & Freymueller, J. (2018). Complex faulting and triggered rupture during the 2018 Mw 7.9 offshore Kodiak, Alaska, earthquake. *Geophysical Research Letters*, 45(15), 7533–7541. https://doi.org/10.1029/ 2018gl078931
- Russo, R. (2011). Western Idaho Shear Zone Passive [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10. 7914/SN/XT_2011
- Satriano, C., Kiraly, E., Bernard, P., & Vilotte, J.-P. (2012). The 2012 Mw 8.6 Sumatra earthquake: Evidence of westward sequential seismic ruptures associated to the reactivation of a N-S ocean fabric. *Geophysical Research Letters*, 39(15), L15302. https://doi.org/10.1029/ 2012gl052387
- Schurr, B., Asch, G., Hainzl, S., Bedford, J., Hoechner, A., Palo, M., et al. (2014). Gradual unlocking of plate boundary controlled initiation of the 2014 Iquique earthquake. *Nature*, 512(7514), 299–302. https://doi.org/10.1038/nature13681
- Schurr, B., Moreno, M., Tréhu, A. M., Bedford, J., Kummerow, J., Li, S., & Oncken, O. (2020). Forming a Mogi doughnut in the years prior to and immediately before the 2014 M8.1 Iquique, northern Chile, earthquake. *Geophysical Research Letters*, 47(16), e2020GL088351. https://doi. org/10.1029/2020gl088351
- Simon Klemperer, K. M. (2010). Collaborative Research: 4D multi-disciplinary investigation of highly variable crustal response to continental extension in the north-central Basin and Range [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/ SN/YX_2010
- Simons, M., Minson, S. E., Sladen, A., Ortega, F., Jiang, J., Owen, S. E., et al. (2011). The 2011 magnitude 9.0 Tohoku-Oki earthquake: Mosaicking the megathrust from seconds to centuries. *Science*, 332(6036), 1421–1425. https://doi.org/10.1126/science.1206731
- Sippl, C., Moreno, M., & Benavente, R. (2021). Microseismicity appears to outline highly coupled regions on the Central Chile megathrust. Journal of Geophysical Research: Solid Earth, 126(11), e2021JB022252. https://doi.org/10.1029/2021jb022252
- Sippl, C., Schurr, B., Asch, G., & Kummerow, J. (2018). Seismicity structure of the Northern Chile forearc from > 100,000 double-difference relocated hypocenters. *Journal of Geophysical Research: Solid Earth*, 123(5), 4063–4087. https://doi.org/10.1002/2017jb015384
- Socquet, A., Hollingsworth, J., Pathier, E., & Bouchon, M. (2019). Evidence of supershear during the 2018 magnitude 7.5 Palu earthquake from space geodesy. *Nature Geoscience*, 12(3), 192–199. https://doi.org/10.1038/s41561-018-0296-0
- Spudich, P., & Cranswick, E. (1984). Direct observation of rupture propagation during the 1979 Imperial Valley earthquake using a short baseline accelerometer array. *Bulletin of the Seismological Society of America*, 74(6), 2083–2114. https://doi.org/10.1785/bssa0740062083
- Steve Gao, K. L. (2012). Passive seismic study of early rifting in Botswana, Zambia, and Malawi [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/XK_2012
- Strollo, A., Cambaz, D., Clinton, J., Danecek, P., Evangelidis, C. P., Marmureanu, A., et al. (2021). EIDA: The European Integrated Data Archive and Service Infrastructure within ORFEUS. Seismological Research Letters, 92(3), 1788–1795. https://doi.org/10.1785/0220200413
- Styron, R., & Pagani, M. (2020). The GEM global active faults database. *Earthquake Spectra*, 36(1_suppl), 160–180. https://doi.org/10.1177/8755293020944182
- Susan Beck, G. Z. (2010). Central Andean Uplift and the Geodynamics of the High Topography [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/ZG_2010
- Tadapansawut, T., Okuwaki, R., Yagi, Y., & Yamashita, S. (2021). Rupture process of the 2020 caribbean earthquake along the oriente transform fault, involving supershear rupture and geometric complexity of fault. *Geophysical Research Letters*, 48(1), e2020GL090899. https://doi.org/ 10.1029/2020gl090899
- Taymaz, T., Ganas, A., Yolsal-Çevikbilen, S., Vera, F., Eken, T., Erman, C., et al. (2021). Source mechanism and rupture process of the 24 January 2020 Mw 6.7 Doğanyol–Sivrice earthquake obtained from seismological waveform analysis and space geodetic observations on the East Anatolian Fault Zone (Turkey). *Tectonophysics*, 804, 228745. https://doi.org/10.1016/j.tecto.2021.228745
- Tian, D. (2020). seisman/HinetPy: 0.6.6 [Software]. Zenodo. https://doi.org/10.5281/zenodo.3695076
- Tilmann, F., Zhang, Y., Moreno, M., Saul, J., Eckelmann, F., Palo, M., et al. (2016). The 2015 Illapel earthquake, central Chile: A type case for a characteristic earthquake? *Geophysical Research Letters*, 43(2), 574–583. https://doi.org/10.1002/2015gl066963
- Twardzik, C., & Ji, C. (2015). The Mw 7.9 2014 intraplate intermediate-depth Rat Islands earthquake and its relation to regional tectonics. Earth and Planetary Science Letters, 431, 26–35. https://doi.org/10.1016/j.epsl.2015.08.033
- UC San Diego. (2013). Central and Eastern US Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/N4
- Universidad De Chile. (2013). Red Sismologica Nacional [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/C1
- University of Genoa. (1967). Regional Seismic Network of North Western Italy [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/GU
- University of Oregon. (1990). University of Oregon Regional Network & Pacific Northwest Seismic Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/UO
- University of Puerto Rico. (1986). Puerto Rico Seismic Network (PRSN) & Puerto Rico Strong Motion Program (PRSMP) [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/PR
- University of Utah. (1962). University of Utah Regional Seismic Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/UU
- University of Washington. (1963). Pacific Northwest Seismic Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/UW
- Utrecht University (UU Netherlands). (1983). NARS [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/NR
- Vallée, M., & Doeut, V. (2016). A new database of source time functions (STFs) extracted from the SCARDEC method. *Physics of the Earth and Planetary Interiors*, 257, 149–157. https://doi.org/10.1016/j.pepi.2016.05.012
- Vera, F., Tilmann, F., & Saul, J. (2024). A decade of short-period earthquake rupture histories from multi-array back-projection. V. 3.0 [Dataset]. GFZ Data Services. https://doi.org/10.5880/GFZ.2.4.2024.001
- Vera, F., Tilmann, F., Saul, J., & Evangelidis, C. P. (2023). Imaging the 2007 Mw 7.7 Tocopilla earthquake from short-period back-projection. Journal of South American Earth Sciences, 127, 104399. https://doi.org/10.1016/j.jsames.2023.104399



- Vernon, F., & BenZion, Y. (2010). San Jacinto Fault Zone Experiment [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/YN_2010
- Wagner, L. (2012). Pre-Hydrofracking Regional Assessment of Central Carolina Seismicity [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/XQ_2012
- Walker, K. T., Ishii, M., & Shearer, P. M. (2005). Rupture details of the 28 March 2005 Sumatra Mw 8.6 earthquake imaged with teleseismic P waves. *Geophysical Research Letters*, 32(24), L24303. https://doi.org/10.1029/2005gl024395
- Walker, K. T., & Shearer, P. M. (2009). Illuminating the near-sonic rupture velocities of the intracontinental Kokoxili Mw 7.8 and Denali fault Mw 7.9 strike-slip earthquakes with global P wave back projection imaging. *Journal of Geophysical Research*, 114(B2), B02304. https://doi. org/10.1029/2008jb005738
- Wang, D., Mori, J., & Koketsu, K. (2016). Fast rupture propagation for large strike-slip earthquakes. Earth and Planetary Science Letters, 440, 115–126. https://doi.org/10.1016/j.epsl.2016.02.022
- Wang, K., Huang, T., Tilmann, F., Peacock, S. M., & Lange, D. (2020). Role of serpentinized mantle wedge in affecting megathrust seismogenic behavior in the area of the 2010 M = 8.8 Maule earthquake. *Geophysical Research Letters*, 47(22), e2020GL090482. https://doi.org/10.1029/ 2020gl090482
- Wells, D. L., & Coppersmith, K. J. (1994). New empirical relationships among magnitude, rupture length, rupture width, rupture area, and surface displacement. Bulletin of the Seismological Society of America, 84(4), 974–1002. https://doi.org/10.1785/bssa0840040974
- William Menke, V. L. (2012). Deep Structure of Three Continental Sutures in Eastern North America [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/X8_2012
- Xie, Y., & Meng, L. (2020). A multi-array back-projection approach for tsunami warning. *Geophysical Research Letters*, 47(14), e2019GL085763. https://doi.org/10.1029/2019gl085763
- Yagi, Y., & Okuwaki, R. (2015). Integrated seismic source model of the 2015 Gorkha, Nepal, earthquake. *Geophysical Research Letters*, 42(15), 6229–6235. https://doi.org/10.1002/2015gl064995
- Yao, H., Gerstoft, P., Shearer, P. M., & Mecklenbräuker, C. (2011). Compressive sensing of the Tohoku-Oki Mw 9.0 earthquake: Frequencydependent rupture modes. *Geophysical Research Letters*, 38(20), L20310. https://doi.org/10.1029/2011gl049223
- Yao, H., Shearer, P. M., & Gerstoft, P. (2013). Compressive sensing of frequency-dependent seismic radiation from subduction zone megathrust ruptures. Proceedings of the National Academy of Sciences, 110(12), 4512–4517. https://doi.org/10.1073/pnas.1212790110
- Ye, L., Lay, T., & Kanamori, H. (2014). The 23 June 2014 Mw 7.9 Rat Islands archipelago, Alaska, intermediate depth earthquake. *Geophysical Research Letters*, 41(18), 6389–6395. https://doi.org/10.1002/2014gl061153
- Ye, L., Lay, T., & Kanamori, H. (2020). Anomalously low aftershock productivity of the 2019 Mw 8.0 energetic intermediate-depth faulting beneath Peru. Earth and Planetary Science Letters, 549, 116528. https://doi.org/10.1016/j.epsl.2020.116528
- Ye, L., Lay, T., & Kanamori, H. (2021). The 25 march 2020 M_w 7.5 Paramushir, northern Kuril Islands earthquake and major ($M_w \ge 7.0$) near-trench intraplate compressional faulting. *Earth and Planetary Science Letters*, 556, 116728. https://doi.org/10.1016/j.epsl.2020.116728
- Yin, J., Yang, H., Yao, H., & Weng, H. (2016). Coseismic radiation and stress drop during the 2015 Mw 8.3 Illapel, Chile megathrust earthquake. Geophysical Research Letters, 43(4), 1520–1528. https://doi.org/10.1002/2015gl067381
- Yue, H., Lay, T., Freymueller, J. T., Ding, K., Rivera, L., Ruppert, N. A., & Koper, K. D. (2013). Supershear rupture of the 5 January 2013 Craig, Alaska (Mw 7.5) earthquake. Journal of Geophysical Research: Solid Earth, 118(11), 5903–5919. https://doi.org/10.1002/2013jb010594
- ZAMG-Zentralanstalt Für Meterologie Und Geodynamik. (1987). Austrian Seismic Network [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/OE
- Zeng, H., Wei, S., & Wu, W. (2020). Sources of uncertainties and artefacts in back-projection results. *Geophysical Journal International*, 220(2), 876–891.
- Zhao, D., Huang, Z., Umino, N., Hasegawa, A., & Kanamori, H. (2011). Structural heterogeneity in the megathrust zone and mechanism of the 2011 Tohoku-oki earthquake (Mw 9.0). *Geophysical Research Letters*, 38(17), L17308. https://doi.org/10.1029/2011g1048408